# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 8-K

Current Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): October 08, 2020

# **Medley Capital Corporation**

(Exact Name of Registrant as Specified in its Charter)

#### **Delaware**

1-35040 (Commission File Number) (State or other jurisdiction of incorporation)

27-4576073 (I.R.S. Employer Identification No.)

280 Park Avenue, 6<sup>th</sup> Floor East New York, NY 10017

(Address of Principal Executive Offices and Zip Code)

Registrant's telephone number, including area code: (212) 759-0777

#### **Not Applicable**

(Former Name or Former Address, if Changed Since Last Report)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$0.001 per share	MCC	The New York Stock Exchange
6.500% Notes due 2021	MCX	The New York Stock Exchange
6.125% Notes due 2023	MCV	The New York Stock Exchange

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))
	cate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (17 CFR §230.405) or Rule 12b-2 e Securities Exchange Act of 1934 (17 CFR §240.12b-2).
Eme	rging growth company $\square$
	emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised accounting standards provided pursuant to Section 13(a) of the Exchange Act.

#### Item 1.01 Entry into a Material Definitive Agreement.

Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

On October 8, 2020, Medley Capital Corporation ("MCC"), Great American Life Insurance Company ("GALIC"), MCC Senior Loan Strategy JV I LLC (the "MCC JV"), and an affiliate of Golub Capital LLC ("Golub") entered into a Membership Interest Purchase Agreement (the "Agreement") pursuant to which a fund affiliated with and managed by Golub concurrently purchased all of MCC's interest in the MCC JV and all of GALIC's interest in the MCC JV for a pre-adjusted gross purchase price of \$156.4 million and an adjusted gross purchase price (which constitutes the aggregate consideration for the membership interests) of \$145.3 million (giving effect to adjustments primarily for principal and interest payments from portfolio companies of MCC JV from July 1, 2020 through October 7, 2020), resulting in net proceeds (before transaction expenses) of \$41.0 million and \$6.6 million for MCC and GALIC, respectively, on the terms and subject to the conditions set forth in the Agreement, including the representations, warranties, covenants and indemnities contained therein. MCC estimates that transaction expenses will be approximately \$1.6 million resulting in net proceeds (including estimated transaction expenses) of \$39.5 million. In connection with the closing of the transaction on October 8, 2020, MCC JV repaid in full all outstanding borrowings under, and terminated, its senior secured revolving credit facility, dated as of August 4, 2015, as amended, administered by Deutsche Bank AG, New York Branch.

The foregoing description of the Agreement is not complete and is qualified in its entirety by reference to a copy of the Agreement, which is filed as Exhibit 10.1 to this Current Report on Form 8-K and incorporated by reference herein.

#### Item 2.01 Completion of Acquisition or Disposition of Assets

The information set forth in Item 1.01 of this Current Report on Form 8-K is incorporated by reference into this Item 2.01.

**Item 7.01 Regulation FD Disclosure** 

On October 9, 2020, MCC issued a press release announcing the entry into the Agreement. The press release is furnished herewith as Exhibit 99.1.

The information in Item 7.01 of this Current Report on Form 8-K, including Exhibit 99.1 furnished herewith, is being furnished and shall not be deemed "filed" for any purpose of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of such Section. The information in this Current Report on Form 8-K shall not be deemed to be incorporated by reference into any filing under the Securities Act of 1933, as amended, or the Exchange Act, except as shall be expressly set forth by specific reference in such filing.

#### **Forward-Looking Statements**

This Current Report contains "forward-looking statements" within the meaning of Section 21E of the Exchange Act. Such statements include, but are not limited to, statements regarding the ongoing obligations under the Agreement, including the indemnities contained in the Agreement, and the possibility that MCC may explore strategic alternatives and other statements containing the words "believes," "anticipates," "plans," "expects," "will" and similar expressions. In addition, the estimated transaction expenses and the unaudited pro forma condensed consolidated financial information giving effect to the disposition of the MCC JV and the related notes thereto, which are attached to this Current Report, include certain assumptions as described therein, including, but not limited to, assumptions with respect to estimates of transaction expenses incurred by the Company and the resulting net proceeds received by the Company. Such forward-looking statements represent management's current expectations and are inherently uncertain. There are a number of important factors that could materially impact the value of the MCC's common stock or cause actual results to differ materially from those indicated by such forward-looking statements. These important factors include, but are not limited to, potential losses of MCC in connection with its indemnification obligations under the Agreement; MCC's ability to comply with the covenants and conditions under its debt instruments; the effects of changes in MCC's credit rating; MCC's ability to arrange and consummate financing or sale transactions or to access capital; whether MCC is able to generate sufficient cash flows and maintain adequate liquidity to meet its liquidity needs, service its indebtedness, repay existing debt obligations, and finance the ongoing obligations of its business; uncertainties associated with the impact from the COVID-19 pandemic, including its impact on the global and U.S. capital markets, the global and U.S. economy, the operational and financial performance of MCC's portfolio companies, and liquidity, and the important factors discussed under the caption "Risk Factors" in Part 1. Item 1A of MCC's Form 10-K for the fiscal year ended September 30, 2019, and its other reports filed with the Securities and Exchange Commission. These important factors, among others, could cause actual results to differ materially from those indicated by forward-looking statements made herein and presented elsewhere by management from time to time. Any such forward-looking statements represent management's estimates as of the date of this Current Report. While MCC may elect to update such forward-looking statements at some point in the future, MCC disclaims any obligation to do so, even if subsequent events cause its views to change. These forward-looking statements should not be relied upon as representing MCC's views as of any date subsequent to the date of this Current Report.

#### Item 9.01 Financial Statements and Exhibits.

- (a) Not Applicable.
- (b) Pro Forma Financial Information.

Unaudited pro forma condensed consolidated financial information of MCC giving effect to the disposition of MCC JV, and the related notes thereto, required by Article 11 of Regulation S-X is attached hereto as Exhibit 99.2.

(c) Not Applicable.

# (d) Exhibits.

Exhibit No.	Description
10.1	Membership Interest Purchase Agreement, dated as of October 8, 2020, by and among Medley Capital Corporation, Great American Life Insurance Company, MCC Senior Loan Strategy JV I LLC and GEMS Fund 5, L.P.
99.1	Press Release dated as of October 9, 2020.
99.2	<u>Unaudited Pro Forma Condensed Consolidated Financial Information.</u>

# **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: October 13, 2020 MEDLEY CAPITAL CORPORATION

By: /s/ Richard T. Allorto, Jr.

Name: Richard T. Allorto, Jr. Title: Chief Financial Officer

#### Medley Capital Corporation Sells the MCC Senior Loan Strategy JV Portfolio to Fund Managed by Golub Capital LLC

New York, NY (October 9, 2020) - The Special Committee of the Board of Directors of Medley Capital Corporation (NYSE: MCC) ("MCC") is pleased to announce that on October 8, 2020, MCC, MCC Senior Loan Strategy JV I LLC (the "MCC JV"), the other holder of membership interests in the MCC JV (the "Other MCC JV Member"), and an affiliate of Golub Capital LLC ("Golub") entered into a Membership Interest Purchase Agreement ("the Agreement") pursuant to which a private fund affiliated with and managed by Golub concurrently purchased all of MCC's interest in the MCC JV and all of the Other MCC JV Member's interest in the MCC JV for a total purchase price of \$156.4 million (subject to various adjustments), on terms and conditions set forth in the Agreement. In connection with the closing of the transaction, MCC JV repaid in full all outstanding borrowings under, and terminated, its senior secured revolving credit facility, dated as of August 4, 2015, as amended, administered by Deutsche Bank AG, New York Branch. Taking into account the debt repayment and the adjustments, MCC and the Other MCC JV Member received net proceeds (before transaction expenses) of \$41.0 million and \$6.6 million, respectively.

"We believe this is a significant transaction for MCC as it improves MCC's liquidity and materially strengthens its balance sheet," said David Lorber, Chair of the MCC Special Committee. He added, "The transaction better positions MCC to meet its other debt repayment obligations."

The Special Committee was advised by Houlihan Lokey for this Transaction.

#### ABOUT MEDLEY CAPITAL CORPORATION

Medley Capital Corporation is a closed-end, externally managed business development company ("BDC") that has common stock which trades on the New York Stock Exchange (NYSE: MCC) and has outstanding bonds which trade on the New York Stock Exchange under the symbols (NYSE: MCV) and (NYSE: MCX). Medley Capital Corporation's investment objective is to generate current income and capital appreciation by lending to privately-held middle market companies, primarily through directly originated transactions, to help these companies expand their businesses, refinance and make acquisitions. Our portfolio generally consists of senior secured first lien loans and senior secured second lien loans. Medley Capital Corporation is externally managed by MCC Advisors LLC, which is an investment adviser registered under the Investment Advisers Act of 1940, as amended. For additional information, please visit Medley Capital Corporation at www.medleycapitalcorp.com.

#### ABOUT MCC ADVISORS LLC

MCC Advisors LLC is a subsidiary of Medley Management Inc. (NYSE: MDLY, "Medley"). Medley is an alternative asset management firm offering yield solutions to retail and institutional investors. Medley's national direct origination franchise is a premier provider of capital to the middle market in the U.S. Medley has \$3.6 billion of assets under management in two business development companies, Medley Capital Corporation (NYSE: MCC) (TASE: MCC) and Sierra Income Corporation, and several private investment vehicles. Over the past 18 years, we have provided capital to over 400 companies across 35 industries in North America. [1] For additional information, please visit Medley Management Inc. at www.mdly.com.

Medley LLC, the operating company of Medley Management Inc., has outstanding bonds which trade on the New York Stock Exchange under the symbols (NYSE:MDLX) and (NYSE:MDLQ).

#### **Forward-Looking Statements**

This press release contains "forward-looking statements" within the meaning of Section 21E of the Securities Exchange Act of 1934, as amended. Such statements include, but are not limited to, statements regarding the ongoing obligations under the Agreement, including the indemnities contained in the Agreement, and the possibility that MCC may explore strategic alternatives and other statements containing the words "believes," "anticipates," "expects," expressions. Such forward-looking statements represent management's current expectations and are inherently uncertain. There are a number of important factors that could materially impact the value of the MCC's common stock or cause actual results to differ materially from those indicated by such forward-looking statements. These important factors include, but are not limited to, potential losses of MCC in connection with its indemnification obligations under the Agreement; MCC's ability to comply with the covenants and conditions under its debt instruments; the effects of changes in MCC's credit rating; MCC's ability to arrange and consummate financing or sale transactions or to access capital; whether MCC is able to generate sufficient cash flows and maintain adequate liquidity to meet its liquidity needs, service its indebtedness, repay existing debt obligations, and finance the ongoing obligations of its business; uncertainties associated with the impact from the COVID-19 pandemic, including its impact on the global and U.S. capital markets, the global and U.S. economy, the operational and financial performance of MCC's portfolio companies, and liquidity; and the important factors discussed under the caption "Risk Factors" in Part 1. Item 1A of MCC's Form 10-K for the fiscal year ended September 30, 2019, and its other reports filed with the Securities and Exchange Commission. These important factors, among others, could cause actual results to differ materially from those indicated by forward-looking statements made herein and presented elsewhere by management from time to time. Any such forward-looking statements represent management's estimates as of the date of this press release. While MCC may elect to update such forward-looking statements at some point in the future, MCC disclaims any obligation to do so, even if subsequent events cause its views to change. These forward-looking statements should not be relied upon as representing MCC's views as of any date subsequent to the date of this press release.

SOURCE: Medley Capital Corporation

Investor Relations Contact: Sam Anderson Head of Capital Markets & Risk Management Medley Management Inc. 212-759-0777

Media Contact: Jonathan Gasthalter/Nathaniel Garnick Gasthalter & Co. LP 212-257-4170

[1] Medley Management Inc. is the parent company of Medley LLC and several registered investment advisors (collectively, "Medley"). Assets under management refers to assets of Medley's funds, which represents the sum of the net asset value of such funds, the drawn and undrawn debt (at the fund level, including amounts subject to restrictions) and uncalled committed capital (including commitments to funds that have yet to commence their investment periods). Assets under management are as of June 30, 2020.

# Medley Capital Corporation Unaudited Pro Forma Condensed Consolidated Financial Statements

On October 8, 2020, Medley Capital Corporation (the "Company"), Great American Life Insurance Company ("GALIC"), MCC Senior Loan Strategy JV I LLC (the "MCC JV"), and an affiliate of Golub Capital LLC ("Golub") entered into a Membership Interest Purchase Agreement (the "Agreement") pursuant to which a fund affiliated with and managed by Golub concurrently purchased all of the Company's interest in the MCC JV and all of GALIC's interest in the MCC JV for a total purchase price of approximately \$156.4 million (subject to various adjustments), resulting in net proceeds (less such adjustments, the credit facility repayments and estimated transaction expenses) of approximately \$39.5 million for the Company, on the terms and subject to the conditions set forth in the Agreement, including the representations, warranties, covenants and indemnities contained therein (the "MCC JV Sale"). In connection with the closing of the transaction on October 8, 2020, MCC JV repaid in full all outstanding borrowings under, and terminated, its senior secured revolving credit facility, dated as of August 4, 2015, as amended, administered by Deutsche Bank AG, New York Branch (the "DB Credit Facility Repayment").

The following unaudited pro forma financial statements of the Company are presented in accordance with Article 11 of Regulation S-X under the Securities Exchange Act of 1934, as amended. The unaudited pro forma condensed consolidated financial statements reflect adjustments to give effect to the MCC JV Sale and indirectly the DB Credit Facility Repayment. In that regard, because the MCC JV constituted an off-balance sheet arrangement with respect to the Company's financial statements the DB Credit Facility Repayment did not directly impact the Company's liabilities on a pro forma basis. However, the DB Credit Facility Repayment did impact the net proceeds received by the Company following the MCC JV Sale. The pro forma adjustments related to the MCC JV Sale are collectively referred to as the "Transaction Adjustments".

The Company's historical balance sheet as of June 30, 2020 has been adjusted to give effect to the Transaction Adjustments as if the related events took place on June 30, 2020. The Company's historical statements of operations for the nine months ended June 30, 2020 and for the fiscal year ended September 30, 2019 have been adjusted to give effect to the Transaction Adjustments as if the transactions were consummated on October 1, 2018.

The unaudited pro forma condensed consolidated financial information are presented for illustrative purposes only and do not necessarily indicate the consolidated results of operations or the financial position that would have resulted had the Transaction Adjustments been completed at the beginning of the applicable period presented, and should not be taken as representative of the Company's future consolidated results of operations or financial position. The pro forma adjustments are based on available information and certain assumptions that the Company believes are reasonable. The unaudited pro forma condensed consolidated financial information should be read in conjunction with the historical consolidated financial statements and related notes contained in the Company's quarterly report Form 10-Q for the fiscal quarter ended June 30, 2020 and annual report on Form 10-K for the fiscal year ended September 30, 2019, both as filed with the Securities and Exchange Commission.

#### Medley Capital Corporation Unaudited Pro Forma Consolidated Statements of Assets and Liabilities As of June 30, 2020 (in thousands, except share and per share data)

	 	_		_	
	Historical		Transaction Adjustments		Pro Forma
ASSETS					
Investments at fair value					
Non-controlled/non-affiliated investments	\$ 116,697			\$	116,697
Affiliated investments	80,257				80,257
Controlled investments	 53,665	_	(47,135) (1)		6,530
Total investments at fair value	250,619		(47,135)		203,484
Cash and cash equivalents	52,203		39,451 (1)		91,654
Other assets	2,042				2,042
Interest receivable	663				663
Receivable for dispositions and investments sold	1,302				1,302
Fees receivable	119				119
Total assets	\$ 306,948	\$	(7,684)	\$	299,264
LIABILITIES					
Notes payable (net of debt issuance costs of \$1,134)	\$ 150,732			\$	150,732
Accounts payable and accrued expenses	4,376				4,376
Interest and fees payable	802				802
Management and incentive fees payable	1,317		(17) (2)		1,317
			17 (3)		
Administrator expenses payable	265				265
Deferred revenue	29				29
Due to affiliate	31				31
Deferred tax liability	50				50
Total liabilities	\$ 157,602	\$	_	\$	157,602
NET ASSETS					
Common stock, par value \$0.001 per share, 100,000,000 common shares authorized, 2,723,711 issued and					
outstanding	\$ 3			\$	3
Capital in excess of par value	673,584				673,584
Total distributable earnings/(loss)	(524,241)		(7,684) (1)		(531,925)
Total net assets	149,346		(7,684)		141,662
Total liabilities and net assets	\$ 306,948	\$	(7,684)	\$	299,264
NET ASSET VALUE PER SHARE	\$ 54.83			\$	52.01

#### Medley Capital Corporation Unaudited Pro Forma Consolidated Statements of Operations For the Nine Months Ended June 30, 2020 (in thousands, except share and per share data)

	Historical		Transaction Adjustments			Pro Forma	
INVESTMENT INCOME							
Interest from investments							
Non-controlled/non-affiliated investments:							
Cash	\$	7,499			\$	7,499	
Payment-in-kind		465				465	
Affiliated investments:							
Cash		691				691	
Payment-in-kind		2,141				2,141	
Controlled investments:							
Cash		85				85	
Payment-in-kind		501				501	
Total interest income		11,382		_		11,382	
Dividend income		4,725		(4,725) <sup>(4)</sup>		_	
Interest from cash and cash equivalents		377				377	
Fee income		618				618	
Total investment income		17,102		(4,725)	_	12,377	
EXPENSES							
Base management fees		4,967		(330) (5)		4,637	
Interest and financing expenses		12,312				12,312	
General and administrative		3,140				3,140	
Administrator expenses		1,743				1,743	
Insurance		988				988	
Directors fees		960				960	
Professional fees, net		(4,797)				(4,797)	
Expenses before expense support reimbursement		19,313		(330)		18,983	
Expense support reimbursement		(349)		22 (6)		(327)	
Total expenses net of expense support reimbursement		18,964		(308)		18,656	
NET INVESTMENT INCOME		(1,862)		(4,417)		(6,279)	
REALIZED AND UNREALIZED GAIN/(LOSS) ON INVESTMENTS							
Net realized gain/(loss) from investments							
Non-controlled/non-affiliated investments		(690)				(690)	
Affiliated investments		_				_	
Controlled investments		(39,076)				(39,076)	
Net realized gain/(loss) from investments		(39,766)		_		(39,766)	
Net unrealized appreciation/(depreciation) on investments							
Non-controlled/non-affiliated investments		(8,423)				(8,423)	
Affiliated investments		2,558				2,558	
Controlled investments		(17,062)		24,126 <sup>(7)</sup>		7,064	
Net unrealized appreciation/(depreciation) on investments		(22,927)		24,126		1,199	
Change in provision for deferred taxes on unrealized (appreciation)/depreciation on investments		(50)				(50)	
Net loss on extinguishment of debt		(2,481)				(2,481)	
Net realized and unrealized gain/(loss) on investments		(65,224)		24,126		(41,098)	
NET INCREASE/(DECREASE) IN NET ASSETS RESULTING FROM OPERATIONS	\$	(67,086)	\$	19,709	\$	(47,377)	
WEIGHTED AVERAGE - BASIC AND DILUTED EARNINGS PER COMMON SHARE	\$	(24.63)	\$	7.24	\$	(17.39)	

WEIGHTED AVERAGE - BASIC AND DILUTED NET INVESTMENT INCOME PER COMMON SHARE	\$	(0.68)	\$ (1.62)	\$ (2.31)
WEIGHTED AVERAGE COMMON STOCK OUTSTANDING - BASIC AND DILUTED	G	2,723,711	2,723,711	2,723,711
DIVIDENDS DECLARED PER COMMON SHARE	\$	_	\$ _	\$ _

#### Medley Capital Corporation Unaudited Pro Forma Consolidated Statements of Operations For the Year Ended September 30, 2019 (in thousands, except share and per share data)

		Historical	Transaction Adjustments		Pro Forma
INTEGENERAL INCOME		_			
INVESTMENT INCOME					
Interest from investments					
Non-controlled/non-affiliated investments:	Φ.	25.260		Φ.	25.200
Cash	\$	25,368		\$	25,368
Payment-in-kind		1,755			1,755
Affiliated investments:		2.400			2.400
Cash		2,198			2,198
Payment-in-kind		2,604			2,604
Controlled investments:		220			220
Cash		338			338
Payment-in-kind		2,801		_	2,801
Total interest income		35,064	- (0		35,064
Dividend income		8,219	(8,050) (4)		169
Interest from cash and cash equivalents		712			712
Fee income		2,304		_	2,304
Total investment income		46,299	(8,050)	_	38,249
EXPENSES					
Base management fees		11,190	(598) <sup>(5)</sup>		10,592
Interest and financing expenses		24,049			24,049
General and administrative		7,399			7,399
Administrator expenses		3,324			3,324
Insurance		623			623
Directors fees		1,258			1,258
Professional fees, net		19,323			19,323
Total expenses		67,166	(598)		66,568
NET INVESTMENT INCOME		(20,867)	(7,452)		(28,319)
REALIZED AND UNREALIZED GAIN/(LOSS) ON INVESTMENTS					
Net realized gain/(loss) from investments					
Non-controlled/non-affiliated investments		(24,762)			(24,762)
Affiliated investments		(7,671)			(7,671)
Controlled investments		(79,740)	(39,124) (8)		(118,864)
Net realized gain/(loss) from investments		(112,173)	(39,124)		(151,297)
Net unrealized appreciation/(depreciation) on investments					
Non-controlled/non-affiliated investments		20,727			20,727
Affiliated investments		(6,864)			(6,864)
Controlled investments		24,635	(8,218) (7)		16,417
Net unrealized appreciation/(depreciation) on investments		38,498	(8,218)	_	30,280
Change in provision for deferred taxes on unrealized (appreciation)/depreciation on investments		_			_

Net loss on extinguishment of debt		(2,033)		(2,033)
Net realized and unrealized gain/(loss) on investments	,	(75,708)	(47,342)	(123,050)
NET INCREASE/(DECREASE) IN NET ASSETS RESULTING FROM OPERATIONS	\$	(96,575)	\$ (54,794)	\$ (151,369)
WEIGHTED AVERAGE - BASIC AND DILUTED EARNINGS PER COMMON SHARE	\$	(35.46)	\$ (20.12)	\$ (55.57)
WEIGHTED AVERAGE - BASIC AND DILUTED NET INVESTMENT INCOME PER COMMON SHARE	\$	(7.66)	\$ (2.74)	\$ (10.40)
WEIGHTED AVERAGE COMMON STOCK OUTSTANDING - BASIC AND DILUTED		2,723,711	2,723,711	2,723,711
DIVIDENDS DECLARED PER COMMON SHARE	\$	_	\$ _	\$ _

# Notes to Unaudited Pro Forma Condensed Consolidated Financial Statements (Dollars in thousands)

#### **Pro Forma Assumptions:**

- (1) Represents pro forma adjustment for the sale of the Company's equity interests in MCC JV as if it took place on June 30, 2020, which incorporated (a) the elimination of the June 30, 2020 fair value of MCC JV (\$47,135); (b) proceeds of \$41,019 received by MCC on October 8, 2020 less \$1,567 of estimated expenses associated with the MCC JV Sale (\$39,451) and (c) the balance sheet impact of the resulted net realized loss of \$40,686 less the reversal of previously-recognized net unrealized loss of \$32,752 (\$7,684).
- (2) Represents pro forma adjustment for the reduction of base management fees payable to the Company's investment adviser, MCC Advisors LLC ("MCC Advisors") consisting of a reduction in the base management fee payable of \$17. The pro forma reduction in the base management fee was attributable to the net realized loss from the MCC JV Sale, which in turn, decreased the average total assets (on which the management fee is calculated) for the quarter then ended.
- (3) Represents the reversal of a portion of the expense support agreement waiver (the "ESA waiver") booked for the month ended June 30, 2020 as a result of the decrease in management fees payable.
- (4) Represents the elimination of actual dividend income received from MCC JV, assuming the MCC JV Sale took place on October 1, 2018.
- (5) Represents pro forma adjustments for the reduction of base management fee expense incurred to MCC Advisors due to the assumed decrease in the average total assets (on which the management fee is calculated), assuming the MCC JV Sale took place on October 1, 2018.
- (6) Represents pro forma adjustment to the ESA waiver as a result of the reduction of base management fees incurred to MCC Advisors (\$22) during the month ended June 30, 2020, assuming the MCC JV sale took place on October 1, 2018.
- (7) Represents elimination of unrealized appreciation/(depreciation) on the MCC JV, assuming the MCC JV Sale took place on October 1, 2018.
- (8) Represents realized loss incurred assuming the MCC JV Sale took place on October 1, 2018.

# MEMBERSHIP INTEREST PURCHASE AGREEMENT

by and among

MEDLEY CAPITAL CORPORATION,

GREAT AMERICAN LIFE INSURANCE COMPANY,

MCC SENIOR LOAN STRATEGY JV I LLC

and

GEMS FUND 5, L.P.

DATED AS OF OCTOBER 8, 2020

# TABLE OF CONTENTS

		Page
ARTICLE I	PURCHASE AND SALE	1
1.1	Purchase and Sale; Purchase Price.	1
1.2	True Sale	
1.3	Withholding	
ARTICLE II	CLOSING; CLOSING DELIVERIES	
2.1	Closing	
2.2	Closing Deliveries. At the Closing,	2
2.3	Allocation of Purchase Price; Tax Matters	4
ARTICLE II	I REPRESENTATIONS AND WARRANTIES REGARDING THE	
SELLERS		7
3.1	Corporate Organization of Medley.	7
3.2	Authority; No Violation	8
3.3	Consents and Approvals.	8
3.4	Title to Membership Interests	9
3.5	Solvency	9
3.6	Independent Investigation	9
3.7	Representations of GALIC	9
	V REPRESENTATIONS AND WARRANTIES REGARDING THE COMPANIES	10
4.1	Organization of the Company; Subsidiaries.	10
4.2	Capitalization	
4.3	Authority; No Violation	12
4.4	Consents and Approvals.	12
4.5	Financial Statements; Undisclosed Liabilities.	13
4.6	Broker's Fees	14
4.7	Absence of Certain Changes or Events	14
4.8	Legal Proceedings	15
4.9	Company Loans	15
4.10	Company-Owned Equity Interests	16
4.11	Company Assets; Title to Assets.	16
4.12	Compliance with Applicable Law.	17
4.13	Books and Records: Officers and Managers: Bank Accounts	18

	4.14	Taxes	18
	4.15	No Change of Control Payments	20
	4.16	Disclaimers	20
ARTIO	CLE V	REPRESENTATIONS AND WARRANTIES OF BUYER	.21
	5.1	Corporate Organization	21
	5.2	Authority; No Violation	
	5.3	Consents and Approvals	
	5.4	Broker's Fees	
	5.5	No Arrangements with Management or Stockholders	22
	5.6	Status of Buyer	
	5.7	Investigation	
ARTIO	CLE VI	ADDITIONAL AGREEMENTS	.23
	6.1	Additional Agreements	23
	6.2	Further Assurances	23
	6.3	Confidentiality.	
	6.4	Waiver	24
ARTIO	CLE VI	I INDEMNITY	.24
	7.1	Survival	24
	7.2	Indemnification	25
	7.3	Defense of Claims	26
	7.4	Adjustment	27
	7.5	Limitations	27
	7.6	Waiver	29
ARTIO	CLE VI	II DEFINED TERMS	.29
ARTIO	CLE IX	GENERAL PROVISIONS	.38
	9.1	Expenses	38
	9.2	Notices	
	9.3	Interpretation	
	9.4	Counterparts; Delivery by Facsimile or PDF	40
	9.5	Entire Agreement	40
	9.6	Governing Law; Jurisdiction	
	9.7	Waiver of Jury Trial	
	9.8	Publicity	41
	9.9	Assignment; Third Party Beneficiaries	41
	9.10	Remedies.	41

9.11	Amendments; waivers42
9.12	Severability
9.13	Schedules
9.14	Representation by Counsel
	<b>EXHIBITS</b>
Exhibit A	Purchase Price Calculation
Exhibit B	Form of Purchase Price Certificate
Exhibit C	Form of Membership Interest Assignment
Exhibit D	Form of Resignation Letter
Exhibit E	Payoff Letter
	<u>APPENDIX</u>
Appendix A	Payment Information and Pro Rata Portion

#### MEMBERSHIP INTEREST PURCHASE AGREEMENT

This **MEMBERSHIP INTEREST PURCHASE AGREEMENT**, dated as of October 8, 2020 (this "Agreement"), is entered into by and among Medley Capital Corporation, a Delaware corporation ("Medley"), Great American Life Insurance Company, an Ohio corporation ("GALIC") (Medley and GALIC, each, a "Seller" and collectively, the "Sellers"), MCC Senior Loan Strategy JV I LLC, a Delaware limited liability company (the "Company"), and GEMS Fund 5, L.P., a Delaware limited partnership ("Buyer").

#### RECITALS:

**WHEREAS**, Sellers collectively own all of the issued and outstanding membership interests (the "*Membership Interests*") in the Company;

**WHEREAS**, Medley owns 87.5% of the Membership Interests in the Company (the "*Medley Interests*") and GALIC owns 12.5% of the Membership Interests in the Company (the "*GALIC Interests*");

**WHEREAS**, Sellers wish to sell to Buyer, and Buyer wishes to purchase from Sellers, the Membership Interests, subject to the terms and conditions set forth herein; and

**NOW, THEREFORE**, in consideration of the mutual covenants, representations, warranties and agreements contained in this Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties agree as follows:

# ARTICLE I PURCHASE AND SALE.

### 1.1 Purchase and Sale; Purchase Price.

- (a) <u>Purchase and Sale.</u> Upon the terms and subject to the conditions of this Agreement, at the Closing, each Seller shall sell to Buyer, and Buyer shall purchase from each Seller, all of each Seller's right, title and interest in and to its Membership Interests, free and clear of all Liens, for the consideration specified in <u>Section 1.1(b)</u>, as apportioned among the Sellers as set forth on <u>Appendix A</u> hereto (such portion is hereinafter referred to as each Seller's "*Pro Rata Portion*") and as described in Section 1.1(c).
- (b) <u>Purchase Price</u>. The aggregate consideration for the Membership Interests shall be an amount in cash equal to \$ 145,317,978.54 (the "*Purchase Price*"), which consists of the components set forth in, and has been calculated in accordance with, <u>Exhibit A</u> hereto. As of the Closing, a certificate, in the form attached hereto as <u>Exhibit B</u>, executed by the Chief Financial Officer of Medley (the "*Purchase Price Certificate*") has been delivered to Buyer, certifying that Medley's calculation of the Purchase Price is in accordance with the terms of this Agreement, including the calculations set forth on <u>Exhibit A</u> hereto, together with reasonable supporting documentation for each such calculation.

- (c) <u>Closing Payments</u>. The following payments shall be made by Buyer at the Closing:
- (i) The Buyer shall pay to each Seller an amount equal to such Seller's Pro Rata Portion of the Purchase Price as set forth on <u>Appendix A</u> in cash by wire transfer of immediately available funds to the account designated by such Seller on <u>Appendix A</u>; and
- (ii) The Buyer shall repay, or cause to be repaid, on behalf of the Acquired Companies, all amounts necessary to discharge the DB Credit Facility by wire transfer of immediately available funds in accordance with the Payoff Letter.
- 1.2 <u>True Sale</u>. The parties hereto expressly intend that the purchase and sale transaction contemplated by this Agreement shall constitute an absolute conveyance of the Membership Interests to Buyer without recourse. In furtherance of the foregoing, at the Closing, Sellers shall update their respective books and records to reflect the fact that the Membership Interests have been sold and that Sellers no longer retain any ownership interest therein.
- 1.3 <u>Withholding</u>. Buyer and any other applicable withholding agent will be entitled to deduct and withhold from any amounts payable pursuant to this Agreement (and any other agreement entered into in connection with the Contemplated Transactions) any withholding Taxes or other amounts required under the Code or any Applicable Law to be deducted and withheld. To the extent any such amounts are so deducted and withheld and properly paid over to the appropriate Governmental Entity or other appropriate Person, such amounts will be treated for all purposes of this Agreement (and any other agreement entered into in connection with the Contemplated Transactions) as having been paid to the Sellers or any other Person in respect of which such deduction and withholding was made.

# ARTICLE II CLOSING; CLOSING DELIVERIES

- 2.1 <u>Closing</u>. On the terms and subject to the conditions set forth in this Agreement, the closing of the Contemplated Transactions (the "*Closing*") shall take place remotely via the electronic or other exchange of documents and signature pages on the date hereof (the "*Closing Date*") simultaneously with the execution of this Agreement by the parties hereto. At the Closing, the parties hereto shall execute and deliver the items referred to in <u>Section 2.2</u>.
  - 2.2 <u>Closing Deliveries</u>. At the Closing,
    - (a) Medley shall deliver to Buyer the following:
- (i) an instrument of assignment in the form of Exhibit C hereto (each, a "Membership Interest Assignment") executed by Medley with respect to the Medley Interests;
- (ii) all authorizations, consents, filings, approvals and Permits necessary to permit the Sellers to perform the transactions contemplated by the Transaction Documents, including any Consents, in each case, in form and substance reasonably satisfactory to Buyer;

- (iii) evidence that all authorized signatories on accounts, lockboxes and other depositories of funds of the Acquired Companies are only Persons designated by Buyer;
- (iv) a certificate executed by a duly authorized officer of Medley, reasonably satisfactory to Buyer, dated as of the Closing Date, certifying that attached thereto are true and complete copies of (i) the resolutions of the board of directors of Medley and the members of the Company authorizing the execution, delivery and performance of the Transaction Documents to which Medley and the Company are a party and the consummation of the Contemplated Transactions, (ii) a certificate of good standing (or equivalent document) for the Acquired Companies issued by the applicable secretary of state of its jurisdiction or organization as of a recent date, (iii) a copy of the certificate of formation (or equivalent document) of the Acquired Companies, certified by the secretary of state of its jurisdiction of organization, and (iv) a copy of the operating agreement (or equivalent document) of the Acquired Companies, in each case, as amended to date;
- (v) written resignations in the form of <u>Exhibit D</u> of each manager, officer, and member of the board of directors or managers of the Acquired Companies, effective from and after the Closing Date;
  - (vi) a duly completed and executed Form W-9;
- (vii) evidence reasonably satisfactory to Buyer of the termination of all Indebtedness of the Acquired Companies and of the termination and release of all Liens on any assets of the Acquired Companies or the Membership Interests and of the termination of any UCC financing statements, including the payoff letter attached as <a href="Exhibit E">Exhibit E</a> (the "Payoff Letter"), which evidences the payoff and termination of all Indebtedness of the Acquired Companies with respect to the DB Credit Facility;
- (viii) written evidence, in form and substance reasonably satisfactory to Buyer, of the termination of that certain Administrative Services Agreement, dated as of March 27, 2015, by and between Medley Capital LLC, a Delaware limited liability company and the Company, including a waiver of the sixty (60) day written notice requirement set forth in Section 7 therein;
  - (ix) the Purchase Price Certificate; and
- (x) such other documents as may be reasonably required to effect the intentions of the parties, executed by Medley and/or the Company.
  - (b) GALIC shall deliver to Buyer the following:
- (i) a Membership Interest Assignment executed by GALIC with respect to the GALIC Interests; and
  - (ii) a duly completed and executed Form W-9.
- (c) Buyer or, without relieving Buyer of its obligations under this Agreement, its Affiliated designees, shall deliver to the Sellers:

- (i) the Purchase Price payable to the Sellers in accordance with Appendix A to the accounts set forth therein; and
- (ii) such other documents as may be reasonably required to effect the intentions of the parties, executed by Buyer.

# 2.3 Allocation of Purchase Price; Tax Matters.

#### (a) Allocation of Purchase Price.

- (i) The parties hereto shall allocate the Purchase Price (and any other amount or item treated as consideration for U.S. federal income Tax purposes) in accordance with the rules under Section 1060 of the Code and the Treasury Regulations promulgated thereunder, as applicable, pursuant to an allocation schedule (an "Allocation Schedule") to be prepared by Buyer. Buyer shall prepare and deliver to Sellers an initial Allocation Schedule within thirty (30) days after the Closing Date.
- If, within twenty (20) days following the delivery of the initial Allocation Schedule, neither Seller notifies Buyer of its disagreement with the Allocation Schedule, the Allocation Schedule shall be final and binding on all parties. If, within such twenty (20)-day period, either Seller notifies Buyer, in writing, that it disputes any item reflected in the initial Allocation Schedule, Buyer and the Sellers shall use commercially reasonable efforts to settle the dispute with respect to such comments promptly. If Buyer and Sellers have not resolved such dispute within thirty (30) days of Buyer's receipt of such Seller's comments, Buyer and Sellers shall jointly retain an Independent Accountant to resolve disputed items in accordance with this Agreement and, absent fraud or manifest error, any determination by the Independent Accountant shall be final and binding on the parties. The costs, fees and expenses of the Independent Accountant shall be borne equally by each Seller asserting a dispute and Buyer or by Medley and Buyer in the case of any dispute asserted by Buyer. Buyer and Sellers hereby covenant and agree to (i) be bound by the final Allocation Schedule for all income Tax purposes, (ii) prepare and file all Tax Returns on a basis consistent with the final Allocation Schedule and (iii) not take any position on any Tax Return, before any Governmental Entity charged with the collection of any Tax, or in any judicial Proceeding that is in any way inconsistent with the terms of the final Allocation Schedule unless required to do so by Applicable Law.
- (iii) If any indemnification payment is made pursuant to this Agreement or any other adjustment to the Purchase Price for U.S. federal income tax purposes occurs, Buyer shall promptly revise the final Allocation Schedule to take into account such payment or adjustment in a manner consistent with the principles of Section 1060 of the Code and the regulations thereunder (and any corresponding provision of state, local or foreign Tax Law, as appropriate).
- (b) <u>Tax Returns</u>. Medley shall prepare, or cause to be prepared, all income Tax Returns required to be filed by the Acquired Companies after the Closing Date with respect to a Pre-Closing Tax Period (each, a "Seller Prepared Tax Return"). Buyer shall prepare, or cause to be prepared, all other Tax Returns required to be filed by the Acquired Companies after

the Closing Date with respect to a Pre-Closing Tax Period (each, a "Buyer Prepared Tax Return"). All Tax Returns prepared pursuant to this Section 2.3(b) shall be prepared in a manner consistent with past practice unless otherwise required by Applicable Law or this Agreement. The Sellers and Buyer agree that the methodology set forth in Section 2.3(c) will, where applicable, be used for preparing all Tax Returns for Straddle Periods. The preparing party shall deliver any such Tax Return to each non-preparing party for its review and comment (together with schedules, statements and supporting documentation) at least thirty (30) days prior to the due date (including extensions) of such Tax Return (or, if any such Tax Return is due within thirty (30) days of the Closing Date, as soon as reasonably practicable following the Closing Date to permit the non-preparing party a reasonable period of time to review). If a non-preparing party objects to any item on any such Tax Return, such non-preparing party shall, within ten (10) days after delivery of such Tax Return, notify the preparing party, in writing, that it so objects. With respect to a Seller Prepared Tax Return, Sellers shall consider in good faith any objections raised by Buyer within ten (10) days after delivery of such Tax Return to Buyer; provided, however, to the extent that (i) such Seller Prepared Tax Return is not prepared in a manner consistent with past practice (unless otherwise required by Applicable Law or this Agreement) and (ii) any item on such Seller Prepared Tax Return or any Tax position taken with respect to such Seller Prepared Tax Return is reasonably expected to prejudice Buyer's Tax position in respect of any Post-Closing Tax Period, the parties agree to resolve any such dispute in accordance with the procedure set forth in this Section 2.3(b). With respect to a Buyer Prepared Tax Return and a Seller Prepared Tax Return that is under dispute pursuant to the immediately foregoing proviso, (A) if a notice of objection shall be duly delivered, Buyer and the Sellers shall negotiate in good faith and use their reasonable best efforts to resolve such items and (B) if Buyer and the Sellers are unable to reach such agreement within ten (10) days after receipt of such notice, the disputed items shall be resolved by the Independent Accountant in accordance with Section 2.3(a)(ii). The Sellers shall pay to Buyer an amount equal to any Taxes shown to be due and payable on any Tax Return prepared pursuant to this Section 2.3(b) for any Straddle Period, to the extent of the amount that is allocated to the Pre-Closing Tax Period as determined pursuant to Section 2.3(c) upon the later of (x) two (2) days before the due date (taking into account extensions) for the payment of any such Tax and (y) two (2) days following the date upon which the Independent Accountant resolves all applicable disputes with respect thereto. The preparation and filing of any Tax Return of the Acquired Companies that does not relate to a Pre-Closing Tax Period shall be exclusively within the control of Buyer.

- (c) <u>Straddle Period</u>. In the case of Taxes that are payable with respect to a Straddle Period, the portion of any such Taxes that are treated as Pre-Closing Taxes for purposes of this Agreement shall be:
- (i) in the case of Taxes (A) based upon, or related to, income, receipts, profits, wages, capital or net worth; (B) imposed in connection with the sale, transfer or assignment of property or (C) required to be withheld, deemed equal to the amount which would be payable if the Tax year ended with the Closing Date, provided that exemptions, allowances or deductions that are calculated on an annual basis (including, but not limited to, depreciation and amortization deductions) shall be allocated between the period ending on the Closing Date and the period after the Closing Date in proportion to the number of days in each period; and

- (ii) in the case of Taxes other than those specified under clause (i) above, deemed to be the amount of such Taxes for the entire period multiplied by a fraction the numerator of which is the number of days in the period ending on the Closing Date and the denominator of which is the number of days in the entire period.
- (d) <u>Cooperation on Tax Matters</u>. Buyer, the Acquired Companies and the Sellers shall use commercially reasonable efforts to cooperate, as and to the extent reasonably requested by each other, in connection with the preparation and filing of Tax Returns pursuant to this Agreement and any action with respect to Taxes related to Tax periods beginning before the Closing Date. Such cooperation shall include the retention and (upon any other party's request) the provision of records and information that are reasonably relevant to any such Tax Return preparation and any such action, and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.
- Tax Contests. Buyer agrees to give written notice to the Sellers of the receipt of any written notice by any Acquired Company or Buyer which involves the assertion of any claim, or the commencement of any action, relating to Taxes, in respect of which an indemnity may be sought by a Buyer Indemnified Party pursuant to Article VII (a "Tax Claim"), as promptly as is reasonably practicable but in any event no later than ten (10) Business Days after receiving the written notice of such Tax Claim. Failure to provide timely notice to the Sellers pursuant to this Section 2.3(e) shall not affect Buyer Indemnified Parties' right to indemnification hereunder, except to the extent the Sellers are actually prejudiced by such failure. Either of the Sellers, at its election, may control the conduct of such Tax Claim (at the cost of the Sellers); provided, however, that (i) the contest or resolution of any underlying issue of such Tax Claim would not reasonably be expected to have a material and adverse effect on Buyer for any subsequent Tax periods, in which case Buyer (or its agent) shall be entitled to participate in the defense of such Tax Claim and to employ counsel of its choice for such purpose, the fees and expenses of which separate counsel shall be borne solely by Buyer; and (ii) the Sellers shall provide Buyer with a timely and reasonably detailed account of each stage of a Tax Claim in which Buyer does not participate, and obtain the prior written consent of Buyer (which consent shall not be unreasonably conditioned, withheld or delayed) before entering into any settlement of a Tax Claim or ceasing to defend such Tax Claim. In the event (i) that neither Seller elects to control, or (ii) of any action relating to Taxes that would reasonably be expected not to give rise to a Tax Claim, the conduct of any such Tax Claim or action relating to Taxes, as applicable, shall be exclusively within the control of Buyer.
- (f) <u>Transfer Taxes</u>. All Transfer Taxes shall be paid when due fifty percent (50%) by the Sellers and fifty percent (50%) by Buyer. The expense of filing all necessary Tax Returns and other documentation with respect to all such Transfer Taxes shall be similarly borne fifty percent (50%) by Buyer and fifty percent (50%) by the Sellers. Buyer shall file all necessary Tax Returns and other documentation with respect to Transfer Taxes (except to the extent such Tax Returns are required by Applicable Law to be filed by the Sellers), and the Sellers agree to cooperate with Buyer in the filing of any such Tax Returns, including promptly supplying any information in their possession that is reasonably necessary to complete such returns. The parties shall cooperate in good faith to minimize the amount of any such Transfer Taxes to the fullest extent possible under Applicable Laws.

- Membership Interests shall, for U.S. federal, state, and local income Tax purposes, be treated as a purchase and sale of the Membership Interests described in Situation 2 of IRS Revenue Ruling 99-6, 1999-1 C.B. 432, and will be treated from the perspective of the Sellers as a sale of the Membership Interests and from Buyer's perspective as an asset acquisition, and (ii) the Company will terminate for purposes of Section 708 of the Code as a result of the transactions contemplated by this Agreement. The parties agree to file (and to cause their respective Affiliates to file) their U.S. federal income Tax Returns (and state, local or foreign income Tax Returns where applicable) in all respects and for all purposes consistent with such intended treatment, unless otherwise required by Applicable Law. The parties further agree not to take (and to cause their respective Affiliates not to take) any position, whether in any Tax Return, audit, examination, claim, adjustment, litigation or other Proceeding or action with respect to U.S. federal income Tax (and state, local or foreign income Tax where applicable), which is inconsistent with such intended treatment, unless otherwise required by Applicable Law.
- (h) Post-Closing Returns and Elections. Without the prior written consent of Buyer, and subject to Section 2.3(g), unless otherwise required by Applicable Law, the Sellers (and, prior to the Closing, the Company, its Affiliates and their respective Representatives) shall not, to the extent it may affect any Acquired Company, make, change or rescind any Tax election, change or adopt any annual Tax accounting period, change or adopt any method of Tax accounting, file any amended Tax Return or take any position on any Tax Return, file any voluntary disclosure agreement with any Tax authority, enter into any "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax Law), settle or compromise any action relating to Taxes, surrender any right to a refund of Taxes or waive or extend the statute of limitations with respect to any Tax or Tax Return or take any action, or enter into any other transaction, in each case that would have the effect of increasing the Tax Liability of Buyer or the Acquired Companies in respect of any Post-Closing Tax Period.
- (i) <u>Tax Refunds</u>. Any refunds of Taxes of any of the Acquired Companies with respect to any Pre-Closing Tax Period shall be for the account of the Sellers, and Buyer shall pay over to Sellers any such refunds within twenty (20) Business Days after receipt thereof.

# ARTICLE III REPRESENTATIONS AND WARRANTIES REGARDING THE SELLERS

Except as disclosed in the Seller Disclosure Schedule, (a) with respect to Sections 3.1 through 3.6, Medley, solely as to itself and not as to any other Seller, and (b) with respect to Section 3.7, GALIC, solely as to itself and not as to any other Seller, represents and warrants to Buyer as follows:

#### 3.1 Corporate Organization of Medley.

(a) Medley is a corporation duly incorporated, validly existing and in good standing under the laws of the state of its incorporation. Medley has all requisite corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted.

- (b) Medley is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not, individually or in the aggregate, have a Material Adverse Effect on Medley.
- (c) Medley is not in material default under or in material violation of any provision of its Organizational Document.

#### 3.2 Authority; No Violation.

- (a) Medley has full corporate power and authority to execute and deliver the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution and delivery of the Transaction Documents to which it is a party and the consummation of the Contemplated Transactions have been duly and validly authorized and approved by the board of directors of Medley. No other corporate proceedings on the part of Medley are necessary to approve the Contemplated Transactions. This Agreement has been duly and validly executed and delivered by Medley and (assuming due authorization, execution and delivery by Buyer) constitutes the valid and binding obligation of Medley, enforceable against Medley in accordance with its terms, except as may be limited by bankruptcy, insolvency, fraudulent transfer, moratorium, reorganization or similar laws of general applicability relating to or affecting the rights of creditors generally and subject to general principles of equity (the "Enforceability Exception").
- Medley nor the consummation by Medley of the Contemplated Transactions, nor compliance by Medley with any of the terms or provisions of the Transaction Documents to which it is a party, will (i) violate or conflict with any provision of the Organizational Documents of Medley, or (ii) except as would not, individually or in the aggregate, be reasonably expected to be material to Medley (A) violate any Applicable Law or Order applicable to Medley or any of its Subsidiaries, properties or assets, or (B) violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation or imposition of any Lien upon any of the properties or assets of Medley or any of its Subsidiaries under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, franchise, agreement or other Contract, instrument or obligation to which Medley or any of their respective Subsidiaries is a party or by which any of them or any of their respective properties or assets is bound.

#### 3.3 Consents and Approvals.

(a) No consents, qualifications, license, order, authorizations, approvals, or exemptions from, or notices to, or filings or registrations with, any Governmental Entity are necessary in connection with the execution and delivery by Medley of the Transaction Documents to which it is a party or the consummation by Medley of the Contemplated Transactions except for those that have been obtained as of the Closing Date.

- (b) No consents or approvals of, or notices to, any Person (other than a Governmental Entity) are necessary in connection with the execution and delivery by Medley of any Transaction Document to which it is a party or the consummation by Medley of the Contemplated Transactions except for those that have been obtained as of the Closing Date.
- 3.4 <u>Title to Membership Interests</u>. Medley owns, beneficially and of record, the Medley Interests, free and clear of any Lien (other than as provided in the Organizational Documents of the Company provided to Buyer prior to the date hereof). Upon transfer of the Medley Interests to Buyer at the Closing, good and valid title to the Medley Interests will pass to Buyer, free and clear of any Liens.
- 3.5 <u>Solvency</u>. Medley and each of its Subsidiaries is Solvent immediately prior to the Closing and, after giving effect to the Contemplated Transactions, immediately after the Closing will be Solvent. No transfer of property is being made and no obligation is being incurred in connection with the Contemplated Transactions with the intent to hinder, delay or defraud either present or future creditors of Medley or any of its Subsidiaries.
- 3.6 <u>Independent Investigation</u>. Medley acknowledges and agrees that in making its decision to enter into this Agreement and the consummate the Contemplated Transactions, Medley has relied solely upon its own investigation, review and analysis, and except for the representations and warranties of Buyer in <u>Article V</u> and the Transaction Documents, none of Buyer or its respective Representatives makes any representation or warranty, either express or implied.
- 3.7 Representations of GALIC. GALIC is a corporation duly incorporated, validly existing and in good standing under the laws of the state of its incorporation. GALIC has full corporate power and authority to execute and deliver the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution and delivery of the Transaction Documents to which it is a party and the consummation of the Contemplated Transactions have been duly and validly authorized and approved by the board of directors of GALIC. GALIC owns, beneficially and of record, the GALIC Interests, free and clear of any Lien (other than as provided in the Organizational Documents of the Company provided to Buyer prior to the date hereof). Upon transfer of the GALIC Interests to Buyer at the Closing, good and valid title to the GALIC Interests will pass to Buyer, free and clear of any Liens.

# ARTICLE IV REPRESENTATIONS AND WARRANTIES REGARDING THE ACQUIRED COMPANIES

Except as disclosed in the Seller Disclosure Schedule, Medley hereby represents and warrants to Buyer as follows:

# 4.1 Organization of the Company; Subsidiaries.

- (a) The Company is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware. The Company has all requisite limited liability company power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted.
- (b) The Company is duly licensed or qualified to do business as a foreign limited liability company and is in good standing (where such concept is recognized) in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not, individually or in the aggregate, have a Material Adverse Effect on the Company.
- (c) True, complete and correct copies of the Organizational Documents of the Company have previously been made available to Buyer, and the Company is not in default under or in violation of any provision of its Organizational Documents.
- The Company does not own, or has never owned, any Subsidiaries other (d) than (i) MCC JV SPV Funding I LLC, a Delaware limited liability company and direct Subsidiary of the Company ("MCC SPV"), and (ii) MCC JV Investment Holdings LLC, a Delaware limited liability company and direct Subsidiary of MCC SPV ("MCC Holdings"). Except as set forth in the foregoing sentence, neither MCC Holdings nor MCC SPV owns, or has ever owned, any Subsidiaries. 100% of the issued and outstanding Equity Interests of MCC SPV are owned by the Company free and clear of any Liens and 100% of the issued and outstanding Equity Interests of MCC Holdings are owned by MCC SPV free and clear of any Liens, other than Liens granted under the DB Credit Facility and released immediately prior to the Closing. Each of MCC Holdings and MCC SPV (i) is duly organized and validly existing and in good standing under the laws of the State of Delaware, (ii) has the requisite limited liability company or other organizational power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted and (iii) is duly licensed or qualified to do business and is in good standing (where such concept is recognized) in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not, individually or in the aggregate, have a Material Adverse Effect on the Acquired Companies. True, complete and correct copies of the Organizational Documents of MCC Holdings and MCC SPV have previously been made available to Buyer, and neither MCC Holdings nor MCC SPV is in default under or in violation of any provision of its Organizational Documents.

# 4.2 <u>Capitalization</u>.

- (a) The Medley Interests constitute 87.5% of the issued and outstanding Equity Interests of the Company and, assuming the accuracy of the representation of GALIC in Section 3.7, the GALIC Interests constitute 12.5% of the issued and outstanding Equity Interests of the Company, in each case, as of but before giving effect to the Closing. All such Equity Interests (i) have been duly authorized and validly issued, (ii) were not issued in violation of any preemptive or other similar rights, (iii) were not issued in violation of the Organizational Documents of the Company or any other Contract to which a Seller or the Company is a party, and (iv) were issued in compliance with all Applicable Laws. None of the Membership Interests is represented by certificates.
- (b) All of the Equity Interests of MCC Holdings and MCC SPV (i) have been duly authorized and validly issued, (ii) were not issued in violation of any preemptive or other similar rights, (iii) were not issued in violation of the Organizational Documents of such entity or any other Contract to which an Acquired Company is a party, and (iv) were issued in compliance with all Applicable Laws. None of the Equity Interests of MCC Holdings and MCC SPV is represented by certificates.
- (c) Except as set forth in Sections 4.2(a) and (b), there are no issued and outstanding (i) Equity Interests in the Acquired Companies, (ii) securities, bonds, debentures or Indebtedness of an Acquired Company convertible into or exercisable or exchangeable for Equity Interests in an Acquired Company, (iii) options, warrants or other rights or agreements, commitments or understandings of any kind to acquire from an Acquired Company, or other obligations of Sellers or the Acquired Companies to issue, transfer, sell, repurchase or redeem any Equity Interests of or in an Acquired Company, (iv) voting trusts, proxies or other similar agreements or understandings to which Sellers, an Acquired Company is a party or by which Sellers, an Acquired Company is bound with respect to the voting of any Equity Interests in an Acquired Company, or (v) contractual obligations or commitments of any character restricting the transfer of, or requiring the registration for sale of, any Equity Interests in an Acquired Company, except, in the case of the foregoing clauses (iv) and (v), as set forth in the Organizational Documents of the Company. There are no outstanding obligations of an Acquired Company to repurchase, redeem or otherwise acquire any Equity Interests of an Acquired Company.
- (d) No Acquired Company is a guarantor for any Liability or obligation (including Indebtedness) of any other Person and no asset of any Acquired Company is subject to any Lien in respect therefor. None of the Acquired Companies has any Indebtedness outstanding.
- (e) Since its respective formation, none of the Acquired Companies has engaged in any business activities or conducted any operations other than in connection with acquiring and owning the Company Loans and Company-Owned Equity Interests.
- (f) Since October 2, 2020, no Acquired Company has made any dividend, distribution or other payment of cash or other property to any Seller or its Affiliates. The Cash

and Cash Equivalents is an amount no less than the amount of cash and cash equivalents set forth on the Purchase Price Certificate as of October 2, 2020.

#### 4.3 Authority; No Violation.

- (a) The Company has full limited liability company power and authority to execute and deliver the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution and delivery of the Transaction Documents to which it is a party and the consummation of the Contemplated Transactions have been duly and validly authorized and approved by the board of managers of the Company and each of its members. No other limited liability company proceedings on the part of the Company are necessary to approve the Contemplated Transactions. This Agreement has been duly and validly executed and delivered by the Company and (assuming due authorization, execution and delivery by Buyer) constitutes the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the Enforceability Exception.
- (b) Neither the execution and delivery of any Transaction Document by the Company nor the consummation by the Company of the Contemplated Transactions, nor compliance by the Company with any of the terms or provisions of the Transaction Documents to which it is a party, will (i) violate or conflict with any provision of the Organizational Documents of the Company, or (ii) except as would not, individually or in the aggregate, be reasonably expected to be material to the Company (A) violate any Applicable Law or Order applicable to an Acquired Company or any of their respective properties or assets, or (B) violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation or imposition of any Lien upon any of the respective properties or assets of an Acquired Company under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, franchise, agreement or other Contract, instrument or obligation to which an Acquired Company is a party or by which any of them or any of their respective properties or assets is bound.

# 4.4 Consents and Approvals.

- (a) Except as has been obtained by Sellers as of the Closing Date, no consents, qualifications, license, order, authorizations, approvals, or exemptions from, or notices to, or filings or registrations with, any Governmental Entity are necessary in connection with the execution and delivery by the Company of the Transaction Documents or the consummation by the Company of the Contemplated Transactions.
- (b) Except for matters covered in the immediately preceding Section 4.4(a) and for consents required under the DB Credit Facility, no consents or approvals of, or notices to, any Person are necessary in connection with the execution and delivery by the Company of any Transaction Document to which they are a party or the consummation by the Company of the Contemplated Transactions.

# 4.5 Financial Statements; Undisclosed Liabilities.

- Medley has provided to Buyer accurate and complete copies of the Acquired Companies' (i) consolidated unaudited balance sheet as of June 30, 2020, consolidated income statement for the nine-month period ended June 30, 2020, and schedule of investments as of June 30, 2020 (the "Interim Financial Statements") and (ii) schedule of investments and consolidated audited balance sheet as of and income statement for the fiscal year ended September 30, 2019. Except as set forth in Section 4.5 of the Seller Disclosure Schedule, each of the foregoing financial statements (including in all cases the notes thereto, if any) (collectively, the "Financial Statements") (i) have been prepared in accordance with the books and records of the Acquired Companies (which are accurate and complete in all material respects), (ii) fairly present in all material respects the Acquired Companies' consolidated results of operations, cash flows, changes in members' equity and consolidated financial position of the Acquired Companies for the respective fiscal periods or as of the respective dates therein set forth (subject in the case of unaudited statements to recurring year-end audit adjustments normal in nature and amount), and (iii) have been prepared in accordance with GAAP. The Acquired Companies maintain, and have maintained for all periods reflected in the Financial Statements, proper and adequate internal accounting controls that provide assurance that (x) transactions are recorded as necessary to permit accurate preparation of its Financial Statements and to maintain accurate accountability for its assets; (y) the reporting of its assets is compared with existing assets at regular intervals; and (z) accounts, notes and other receivables and payables are recorded accurately, and proper and adequate procedures are implemented to effect the collection thereof on a current and timely basis. None of the Sellers, Acquired Companies or to Medley's Knowledge, any Representative of an Acquired Company has received or otherwise had or obtained knowledge of any material complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, methodologies or methods of an Acquired Company or their respective internal accounting controls, including any material complaint, allegation, assertion or claim that an Acquired Company has engaged in questionable accounting or auditing practices. No independent public accountant of a Seller or an Acquired Company has resigned (or informed such party that it intends to resign) or been dismissed as independent public accountants of a Seller or an Acquired Company as a result of or in connection with any disagreements with either Seller or an Acquired Company on a matter of accounting principles or practices, financial statement disclosure or auditing scope or procedure.
- (b) No Acquired Company has any Liability (whether absolute, accrued, contingent or otherwise and whether due or to become due), except for (i) those Liabilities reflected or reserved against (in accordance with GAAP) in the Interim Financial Statements, (ii) Liabilities incurred in a commercially reasonable manner, and in the ordinary course of business consistent with past practice, since the date of the Interim Financial Statements, which are not, individually or in the aggregate, in excess of \$250,000, none of which results from, arises out of, or relates to any breach or violation of, or default under, a Contract or Applicable Law and (iii) those funding obligations set forth in Section 4.9(a) of the Seller Disclosure Schedule. Notwithstanding the foregoing, no Acquired Company (i) has any Liability to any Person with respect to any management fee or other accrued expense; (ii) is a party to any Contract that will be in effect immediately following the Closing, other than the Company Loan Documents, the Equity Governing Documents, and the Transaction Documents to which the Company is a party;

or (iii) has any Liability to any Person with respect to the matters set forth on <u>Section 4.5(b)(iii)</u> of the Seller Disclosure Schedule.

4.6 <u>Broker's Fees.</u> Except for the retention of and fees payable to Houlihan Lokey Capital, Inc., none of the Sellers or the Acquired Companies has utilized any broker, finder or financial advisor or incurred any liability for any broker's fees, commissions or finder's fees in connection with the Contemplated Transactions.

# 4.7 Absence of Certain Changes or Events.

- (a) Except as set forth on Section 4.7(a) of the Seller Disclosure Schedule and events or circumstances affecting the Syndicated Company Loans, since June 30, 2020, (i) no event or events have occurred that have had or would reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect on the Acquired Companies, (ii) except with respect to the process resulting in the Contemplated Transactions, the Acquired Companies have conducted their business in all material respects only in the ordinary course of business consistent with past practice, and (iii) Medley, the Acquired Companies and their applicable respective Representatives have monitored the Company Loans and the Company-Owned Equity Securities, including any data rooms, correspondence or other materials exchanged in connection therewith, in the ordinary course of business consistent with past practice and have valued the Company Loans and Company-Owned Equity Interests in accordance with their respective valuation policies and procedures.
- (b) Except as set forth on Section 4.7(b) of the Seller Disclosure Schedule, since the execution of that certain Bid Letter, dated as of September 4, 2020, by and between Medley and Buyer, (i) no Acquired Company has acquired, sold, transferred, leased, licensed, pledged, mortgaged, assigned or otherwise disposed of, or encumbered or exchanged, any loans or any portions thereof, (ii) none of Medley or the Acquired Companies have consented to or approved any material modification, amendment, waiver or any other change with respect to any Company Loan, including with respect to any terms thereof, and (iii) there has been no (A) payment or prepayment of a Company Loan in excess of \$150,000 for an individual Company Loan or \$500,000 in the aggregate for all Company Loans (other than scheduled quarterly amortization payments in accordance with the Company Loan Documents), (B) actual or, to Medley's Knowledge, proposed in writing, refinancings of any Company Loan, (C) actual or, to Medley's Knowledge, proposed in writing, incremental financings or upsizes of any Company Loan, (D) actual or, to Medley's Knowledge, proposed in writing, requests for amendments, waivers, consents or changes to any material terms of any Company Loan, and (E) sale, transfer or disposition of any Company Loan.
- (c) Since June 30, 2020, none of the Acquired Companies has, except to the extent expressly contemplated by this Agreement (or any other Transaction Document) taken any action to make, change or rescind any Tax election, changed or adopted any annual Tax accounting period, changed or adopted any method of Tax accounting, filed any amended Tax Return or taken any position on any Tax Return, entered into any "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax Law), settled or compromised any action, examination, audit or investigation relating to Taxes, surrendered any right to a refund of Taxes or waived or extended

the statute of limitations with respect to any Tax or Tax Return (other than an automatic extension to extend the time for filing any Tax Return in the ordinary course of business) or taken any action, or entered into any other transaction, in each case that would have the effect of increasing the Tax Liability of Buyer or the Acquired Companies in respect of any Post-Closing Tax Period.

# 4.8 <u>Legal Proceedings</u>.

- (a) No Acquired Company is a party to any, and there are no (and during the past three (3) years, there have not been any) pending or, to Medley's Knowledge, threatened, material Proceedings against any Acquired Company (or, to Medley's Knowledge, pending or threatened against or affecting any of the Acquired Companies' Representatives with respect to their activities for or on behalf of the Acquired Companies) or to which any of their assets are subject. To Medley's Knowledge, there is no basis for future Proceedings against an Acquired Company or to which any of their assets, including the Company Loans and the Company-Owned Equity Interests, are subject.
- (b) There is no material judgment, settlement agreement, order, injunction, decree or regulatory restriction (other than those of general application that apply to similarly situated companies or their Subsidiaries) imposed upon an Acquired Company or the assets of an Acquired Company.

# 4.9 Company Loans.

- (a) Section 4.9(a) of the Seller Disclosure Schedule sets forth, as of the Closing Date, a true, correct and complete list of (i) each Company Loan owned by the Acquired Companies, (ii) the name of the Borrower of each Company Loan, (iii) the interest rate on each Company Loan, (iv) the maturity date of each Company Loan, (v) the outstanding unpaid principal amount of each Company Loan, (vi) the amount of accrued interest for each Company Loan; (vii) the amount of accrued but unpaid fees or other amounts (other than accrued interest) for each Company Loan; (viii) the currency for each Company Loan; (ix) any undrawn commitments with respect to each Company Loan; and (x) whether such Company Loan is on accrual or non-accrual status. Except as set forth on Section 4.9(a) of the Seller Disclosure Schedule, no Company Loan constitutes a delayed draw term loan.
- (b) Each Company Loan Document to which an Acquired Company is a party constitutes the legal, valid and binding obligation of such Acquired Company and, to the Knowledge of Medley, the applicable Borrower, enforceable against such Acquired Company and, to the Knowledge of Medley, the applicable Borrower, in accordance with their respective terms (subject to the Enforceability Exception). No fact, event or condition exists that constitutes or, after notice or lapse of time or both, will constitute, a breach, violation or default on the part of an Acquired Company under any Company Loan Document. As of the Closing Date, to Medley's Knowledge (i) there is no outstanding "Event of Default" (or similar terms having comparable meanings) under any Company Loan Document or any "Default" (or similar terms having comparable meanings) under any Company Loan Document, (ii) there has been, since September 4, 2020, no other material and adverse change in facts, events or circumstances with respect to a Company Loan (other than any Syndicated Company Loan), (iii) no Company Loan

is more than thirty (30) days delinquent in the payment of interest or principal therein, and (iv) there is no pending payment, pre-payment, refinancing or similar event relating to any Company Loan (other than payments of interest and amortization in accordance with the Company Loan Documents) for which a Borrower has delivered a notice of repayment.

- (c) Except as set forth on Section 4.9(c) of the Seller Disclosure Schedule, complete and correct copies of all the Company Loan Documents in the possession of the Acquired Companies or an Affiliate thereof, including all modifications, amendments and supplements thereto, have been made available to Buyer. Except as set forth in such documents provided to Buyer, the Company Loan Documents and except as set forth on Section 4.9(c) of the Seller Disclosure Schedule, (1) have not been modified in any material respect, satisfied or canceled in whole or in part, or subordinated to any other indebtedness of the applicable Borrower and (2) are not subject to any release or compliance waiver that is currently in effect as to any provision thereof (or, if such release or compliance waiver exists, was made available to Buyer), except for any such release or compliance waiver that is not material to the Acquired Companies. There are no material disputes pending or, to Medley's Knowledge, threatened against any Acquired Company with respect to any Company Loan Document.
- (d) None of the Company Loan Notes has any marks or notations indicating that it has been pledged, assigned or otherwise conveyed to any Person other than an Acquired Company.

#### 4.10 Company-Owned Equity Interests.

- (a) Section 4.10(a) of the Seller Disclosure Schedule sets forth a true, correct and complete list of (i) each Company-Owned Equity Interests of an Existing Portfolio Company; (ii) the issuer of such Company-Owned Equity Interest, and (iii) any undrawn commitments with respect to each Company-Owned Equity Interest.
- (b) No Acquired Company is in breach or under default of its obligations under, or otherwise pursuant to the terms, conditions or provisions of, any of such Equity Governing Documents. There are no material disputes pending or, to Medley's Knowledge, threatened with respect to any Equity Governing Documents.

# 4.11 Company Assets; Title to Assets.

- (a) Except as set forth in Section 4.11(a) of the Seller Disclosure Schedule, to Medley's Knowledge, there are no Proceedings pending in which one of the Borrowers has (i) filed, or consented (by answer or otherwise) to the filing against it, of a petition for relief under any bankruptcy or insolvency law of any jurisdiction, (ii) made an assignment for the benefit of its creditors, (iii) consented to the appointment of a custodian, receiver, trustee, liquidator or other judicial officer with similar power over itself or any substantial part of its property, (iv) been adjudicated by a court to be insolvent, or (v) taken corporate, limited liability company or partnership action for the purpose of authorizing any of the foregoing.
- (b) Except as set forth in Section 4.11(b)(i) of the Seller Disclosure Schedule, the applicable Acquired Company is the sole owner and holder of the Company Loans and the Company-Owned Equity Interests and such Acquired Company has good and marketable title

and all legal and beneficial interest in and to all of the Company Loans and Company-Owned Equity Interests, free and clear of any Liens (but subject to the terms of the Company Loan Documents or the Equity Governing Documents and restrictions on transfer arising under Applicable Law). Except as set forth in Section 4.11(b)(ii) of the Seller Disclosure Schedule, no Company Loan is subject to a participation or other participating or other interest of any nature whatsoever pursuant to which an Acquired Company has participated its interests (or sold a participating or other interest) in such Company Loan.

(c) The only assets owned by the Acquired Companies are the Company Loans, the Company-Owned Equity Interests and the Cash and Cash Equivalents.

# 4.12 Compliance with Applicable Law.

- (a) The Acquired Companies hold all Permits necessary for the lawful conduct of their respective businesses, and have complied in all material respects with and are not in default in any respect under any, any Permit. The Acquired Companies are, and have been during the past three (3) years, in compliance in all material respects with all Applicable Laws. None of Medley or the Acquired Companies have received any notice, whether written or, to the Knowledge of Medley, oral, involving any failure of any Acquired Company or a Company Loan (other than a Syndicated Company Loan) to be in compliance with Applicable Law. During the past three (3) years, no Acquired Company has been subject to any inspection, finding, investigation, penalty, assessment, formal inquiry, audit or any other compliance or enforcement action by a Governmental Entity.
- (b) None of the Acquired Companies, nor any of their respective officers, directors or employees, nor to the Knowledge of Medley, any other Representative acting on the direction of, or to the Knowledge of Medley, on behalf of the Acquired Companies, is currently or has been since January 1, 2015: (A) a Sanctioned Person, (B) organized, resided or located in a Sanctioned Country, (C) engaging or has engaged in any dealings or transactions with any Sanctioned Person or in any Sanctioned Country, to the extent such activities violate applicable Sanctions Laws or Ex-Im Laws, or (D) otherwise in violation of applicable Sanctions Laws, Ex-Im Laws, or U.S. anti-boycott Laws (collectively, "*Trade Control Laws*"). The Acquired Companies are in compliance, in all material respects, with the United States Public Law 107-56, Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism (USA PATRIOT ACT) Act of 2001, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.
- (c) The Acquired Companies have not, and no Representative or other Person acting for or on behalf of an Acquired Company has, in the course of his or her actions for or on behalf of an Acquired Company, directly or indirectly: (i) used any corporate funds for unlawful contributions, gifts, entertainment or other expenses related to political activity or to retain or engage a prospective client; (ii) made any unlawful payments to foreign or domestic government officials from corporate funds; (iii) established or maintained any unlawful or unrecorded fund or corporate monies or other assets; (iv) made any bribe, rebate, payoff, influence payment, kickback or similar payment; or (v) taken any action in violation of the United States Foreign Corrupt Practices Act of 1977, the UK Bribery Act of 2010 or any other anti-corruption or anti-

bribery or anti-money laundering Law of any jurisdiction (collectively, "Anti-Corruption Laws").

(d) Since January 1, 2015, no Acquired Company has (i) received from any Governmental Entity, or any other Person any written notice, inquiry or internal or external allegation, (ii) made any voluntary or involuntary disclosure to a Governmental Entity, or (iii) conducted any internal investigation or audit concerning any actual or potential violation or wrongdoing related to Trade Control Laws or Anti-Corruption Laws.

#### 4.13 Books and Records; Officers and Managers; Bank Accounts.

- (a) The books of account, minute books, equity record books and other records of the Acquired Companies, all of which have been made available to Buyer, are true and complete in all material respects. At the Closing, all such books and records will be in the possession of the Acquired Companies.
- (b) <u>Section 4.14(b)</u> of the Seller Disclosure Schedule lists all officers, managers and/or directors of the Acquired Companies as of immediately prior to the Closing.
- (c) <u>Section 4.14(c)</u> of the Seller Disclosure Schedule sets forth an accurate and complete list of the name and addresses of every bank or other financial institution in which an Acquired Company maintains an account or in which an account is maintained for the benefit of an Acquired Company (whether checking, saving or otherwise), lock box or safe deposit box, and the account numbers and names of Persons having signing authority or other access thereto.

#### 4.14 Taxes.

- (a) Each of the Company Loans and Company Loan Notes is in "registered form" within the meaning of Section 163(f) of the Code for U.S. federal income tax purposes.
- (b) Since formation, the Company has always been classified as a "partnership" under Treasury Regulations Section 301.7701-2(c)(1) and is not, and has never been, a corporation under Treasury Regulations Section 301.7701-2(b) for U.S. federal income tax purposes and income Tax purposes in each applicable state, local and foreign jurisdiction. Each of MCC Holdings and MCC SPV is and has always been since the date of formation an entity classified as disregarded as separate from the Company as described under Treasury Regulations Section 301.7701-2(c)(1) for U.S. federal income tax purposes and income Tax purposes in each applicable state and local jurisdiction.
- (c) Each of the Acquired Companies has timely filed all income and material other Tax Returns required to be filed by it under Applicable Law and has paid or made adequate provisions for all income and other Taxes (whether or not shown on or required to be shown on any Tax Return) due and payable by such Acquired Company. All such Tax Returns have been prepared according to Applicable Law, in good faith, and are true, correct, and complete in all material respects.
- (d) None of the Acquired Companies is required to file any Tax Returns in any jurisdictions other than those set forth on Section 4.15(d) of the Seller Disclosure Schedules.

No written claim has been made by any Governmental Entity in any jurisdiction where the Acquired Companies do not file Tax Returns that it is, or may be, subject to Tax by that jurisdiction.

- (e) There are no actions by any Governmental Entity in respect of or relating to the Taxes of the Acquired Companies that are currently pending, threatened in writing, or being conducted, nor, to Medley's Knowledge, any examinations, audits or investigations relating to the Taxes of the Acquired Companies and no extensions or waivers of statutes of limitations have been given or requested with respect to any Taxes or Tax Returns of the Acquired Companies other than an automatic extension to extend the time for filing any Tax Return in the ordinary course of business. There is no outstanding written claim concerning any Tax Liability of the Acquired Companies that has been claimed by any Governmental Entity.
- (f) Medley has made available to Buyer complete and true copies of all income and all other material Tax Returns, examination, audit or action reports, and statements of deficiencies assessed against or agreed to by the Acquired Companies for Tax periods beginning on or after January 1, 2016.
- (g) Each of the Acquired Companies has complied in all material respects with all Applicable Laws, rules and regulations relating to the payment and withholding of Taxes (including, but not limited to, any withholding on distributions to nonresident partners under Applicable Law, withholding and reporting requirements under Code Sections 1441 through 1446, 3401 through 3406, 6041 and 6049, and similar provisions under any other requirements under Applicable Law) and has, within the times and manner prescribed by Applicable Law, withheld from amounts paid or owing to any employee, independent contractor, creditor, foreign person or other third party and paid over to proper Governmental Entities on a timely basis all amounts required.
- (h) No Liens for Taxes in respect of any of the Acquired Companies exist with respect to any assets or properties of the Acquired Companies except for statutory liens for current Taxes or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings by an Acquired Company and for which appropriate reserves have been established in accordance with GAAP and are included on the balance sheet.
- (i) None of the Acquired Companies has ever been a member of an Affiliated Group that filed or was required to file a consolidated, combined or unitary Tax Return. None of the Acquired Companies has any Liability for Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any corresponding provision of state, local or foreign Law), as transferee or successor, or by contract (other than commercial contracts entered into in the ordinary course not primarily related to Taxes).
- (j) None of the Acquired Companies is party to or bound by any Tax sharing agreements, Tax allocation agreements, or similar agreements the principal purpose of which relates to Taxes (including any advance pricing agreement, closing agreement or other agreement relating to Taxes with any Governmental Entity). No private letter rulings, technical advice

memoranda or similar agreement or rulings have been requested, entered into, or issued by any Governmental Entity with respect to any of the Acquired Companies.

- (k) None of the Acquired Companies requested any extension of time within which to file any Tax Return, which Tax Return has not yet been filed.
- (1) None of the Acquired Companies has been a party to a transaction that is or is substantially similar to a "reportable transaction," within the meaning of Treasury Regulations Section 1.6011-4(b).
- (m) None of the Acquired Companies will be required to include any item of income in, or exclude any item or deduction from, taxable income for any Tax period or portion thereof ending after the Closing Date as a result of any: (i) change in a method of accounting under Section 481 of the Code (or any comparable provision of state, local or foreign Tax Laws), or use of an improper method of accounting, for a Tax period ending on or prior to the Closing Date; (ii) an installment sale or open transaction occurring on or prior to the Closing Date; (iii) prepaid amount or deferred revenue received on or before the Closing Date (other than prepaid amount or deferred revenue received in the ordinary course of business); or (iv) election under Section 108(i) of the Code.
- (n) Each of the Sellers is not a "foreign person" as that term is used in Section 1446(f) of the Code and Treasury Regulations Section 1.1445-2.
- (o) None of the Acquired Companies has been subject to Tax, engaged in business, or had a permanent establishment in any jurisdiction outside the United States.
- 4.15 <u>No Change of Control Payments</u>. No Acquired Company has made nor is obligated to make any payment (including rescission or liquidated damages) to any Person in connection with the Contemplated Transactions or would be obligated to make any such payment in connection with any other change of control transaction involving an Acquired Company.
- 4.16 <u>Disclaimers.</u> NONE OF THE SELLERS, ANY AFFILIATES THEREOF, OR ANY ADVISERS OR REPRESENTATIVES (FINANCIAL, LEGAL OR OTHERWISE), HAVE MADE ANY REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, OF ANY NATURE WHATSOEVER RELATING TO THE SELLER, THE ACQUIRED COMPANIES OR THE BUSINESS OF THE SELLER OR OTHERWISE IN CONNECTION WITH THE CONTEMPLATED TRANSACTIONS, OTHER THAN THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN (I) <u>ARTICLE III</u> HEREOF, (II) ARTICLE IV HEREOF, OR (III) THE PURCHASE PRICE CERTIFICATE.

Without limiting the generality of the foregoing, no Seller nor any of their respective Affiliates or representatives have made, and shall not be deemed to have made, any representations or warranties in the information or materials relating to the Sellers, the Acquired Companies or the business of the Sellers and the Acquired Companies made available to Buyer and its Affiliates, including projections, confidential information memoranda, due diligence materials, data room materials, or in any presentation by management of the Sellers, the Acquired Companies or others in connection with the Contemplated Transactions, and no

statement contained in any of such information or materials or made in any such presentation shall be deemed a representation or warranty hereunder or otherwise or deemed to be relied upon by the Buyer in executing, delivering and performing this Agreement, any other transaction documents and the transactions contemplated hereby and thereby, in each case, other than with respect to any claim for fraud, willful misconduct or intentional misrepresentation.

# ARTICLE V REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to the Sellers as follows:

5.1 <u>Corporate Organization</u>. Buyer is a limited partnership duly organized, validly existing and in good standing under the laws of the State of Delaware. Buyer has the requisite partnership power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not, individually or in the aggregate, have a Material Adverse Effect on Buyer.

#### 5.2 Authority; No Violation.

- (a) Buyer has full partnership power and authority to execute and deliver the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution and delivery of the Transaction Documents to which it is a party and the consummation of the Contemplated Transactions have been duly and validly approved by Buyer and no other proceedings on the part of Buyer are necessary to approve this Agreement or to consummate the Contemplated Transactions. This Agreement has been duly and validly executed and delivered by Buyer and (assuming due authorization, execution and delivery by the Sellers and the Company) constitutes the valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms (subject to the Enforceability Exception).
- (b) Neither the execution and delivery of the Transaction Documents to which it is a party by Buyer nor the consummation by Buyer of the Contemplated Transactions, nor compliance by Buyer with any of the terms or provisions of this Agreement, will (i) violate any provision of the Organizational Documents of Buyer, or (ii) except as would not, individually or in the aggregate, be reasonably expected to materially adversely affect Buyer's ability to consummate the Contemplated Transactions (A) violate any Applicable Law applicable to Buyer or any of its Subsidiaries, properties or assets, or (B) except as would not, individually or in the aggregate, have a Material Adverse Effect on Buyer, violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation of any Lien upon any of the respective properties or assets of Buyer or any of its Subsidiaries under, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust, license, lease, franchise, agreement or other Contract, instrument or

obligation to which Buyer or any of its Subsidiaries is a party or by which any of them or any of their respective properties or assets is bound.

- 5.3 <u>Consents and Approvals</u>. No consents or approvals of any Person are necessary in connection with the execution and delivery by Buyer of this Agreement or the consummation by Buyer of the Contemplated Transactions except for those that have been obtained as of the Closing Date.
- 5.4 <u>Broker's Fees</u>. Neither Buyer nor any of its Subsidiaries nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any broker's fees, commissions or finder's fees in connection with the Contemplated Transactions.
- 5.5 No Arrangements with Management or Stockholders. Other than this Agreement, as of the Closing Date, there are no contracts, undertakings, commitments, agreements or obligations or understandings, whether written or oral, in each case that are binding, between Buyer or any of its Affiliates, on the one hand, and any member of either Seller's management or the governing bodies, or any stockholder of either Seller, on the other hand, relating to the Contemplated Transactions or the operations of the Acquired Companies or either Seller following Closing.
- 5.6 <u>Status of Buyer</u>. Buyer (a) is a "sophisticated" investor and/or an "accredited" investor as that term is defined in Rule 501 of Regulation D under the Securities Act, (b) is able to bear the economic risk associated with the purchase of the Membership Interests, (c) has such knowledge and experience so as to be aware of the risks and uncertainties inherent in the purchase of rights of the type contemplated in this Agreement, and (d) has independently and without reliance upon either Seller, and based upon such information as Buyer has deemed appropriate, made its own analysis and decision to enter into this Agreement. Buyer is not purchasing the Membership Interests with a view towards sale or distribution thereof in violation of the Securities Act.
- Investigation. Buyer has conducted its own independent review and analysis of 5.7 the businesses, assets, condition, operations and prospects of the Acquired Companies and has been provided access to the properties, premises and records of the Acquired Companies for this purpose. In entering into this Agreement, Buyer has relied solely upon its own investigation and analysis, and Buyer acknowledges that, except for the representations and warranties of the Sellers in Articles III and IV, the Purchase Price Certificate, and the Transaction Documents, none of the Sellers or their respective Subsidiaries nor any of their respective Representatives makes any representation or warranty, either express or implied, as to the accuracy or completeness of any of the information provided or made available to Buyer or its Representatives. Without limiting the generality of the foregoing, none of the Sellers or any of their respective Subsidiaries nor any of their respective Representatives or any other Person has made a representation or warranty to Buyer with respect to (a) any projections, estimates or budgets for the Acquired Companies or (b) any material, documents or information relating to the Acquired Companies made available to Buyer or its Representatives in any "data room," confidential information memorandum or otherwise, except as covered by the Transaction Documents, the Purchase Price Certificate or a representation or warranty in Article III or IV.

Notwithstanding anything to the contrary herein, this <u>Section 5.7</u> shall not apply with respect to any claims for fraud, willful misconduct or intentional misrepresentation that Buyer may have.

#### ARTICLE VI ADDITIONAL AGREEMENTS

- 6.1 <u>Additional Agreements</u>. In case at any time after the Closing any further action is necessary or desirable to carry out the purposes of this Agreement, the proper officers and directors of each Seller shall take all such necessary action as may be reasonably requested by Buyer.
- 6.2 Further Assurances. Each Seller shall, at any time from and after the Closing, upon the reasonable request of Buyer, do, execute, acknowledge and deliver, and cause to be done, executed, acknowledged and delivered, all such further acts, deeds, assignments, transfers, conveyances, powers of attorney or assurances as may be reasonably required to transfer, convey, grant and confirm to and vest in Buyer good title to all of the Membership Interests, free and clear of all Liens, and otherwise consummate the transactions contemplated by this Agreement and the other Transaction Documents. Each of the parties hereto shall, at any time from and after the Closing, upon the reasonable request of any other party hereto, do, execute, acknowledge and deliver, and cause to be done, executed, acknowledged and delivered, all such further acts, deeds, assignments, transfers, conveyances, powers of attorney or assurances as may be reasonably required to consummate the transactions contemplated by this Agreement and the other Transaction Documents.

#### 6.3 Confidentiality.

- (a) The parties agree that the Confidentiality Agreement shall terminate automatically as of the Closing without any further action by the parties and without further obligations or liabilities of Buyer.
- For a period of two (2) years following the Closing, each Seller shall, and will cause each of their respective Subsidiaries, successor, advisor, and their respective Representatives to, treat and hold as strictly confidential any data or information related to the Acquired Companies, Buyer or its direct or indirect shareholders, including information and documents disclosed to either Seller (or any of their respective Representatives), whether before or after the date hereof, pursuant to this Agreement or in connection with the Contemplated Transactions; provided, that Sellers shall not be bound by the confidentiality requirements of this Section 6.3(b) with respect to data or information and documents which (A) are or become generally available to the public other than as a result of a disclosure by either Seller after the Closing Date in breach of its obligations hereunder, (B) are required to be disclosed in connection with the ordinary course of any Seller's or its Representative's SEC's reporting or other regulatory or self-regulatory process, (C) are required to be disclosed by Applicable Law or the rules of any self-regulatory organization so long as reasonable prior notice is given to Buyer of such disclosure and a reasonable opportunity is afforded to Buyer to contest the same, (D) are disclosed in connection with the enforcement of any right or remedy relating to this Agreement or (E) are obtained independently from a source that is authorized to disclosure such data, information and documents.

- For a period of two (2) years following the Closing, Buyer shall, and will cause its Subsidiaries, successor, advisor, and their respective Representatives to, treat and hold as strictly confidential any data or information related to the Sellers and their respective Subsidiaries, or its direct or indirect shareholders, including information and documents disclosed to Buyer (or its respective Representatives), whether before or after the date hereof, pursuant to this Agreement or in connection with the Contemplated Transactions; provided, that Buyer shall not be bound by the confidentiality requirements of this Section 6.3(c) with respect to data or information and documents which (A) relates to the Company Loans, Company-Owned Equity Securities, or the Acquired Companies, (B) are or become generally available to the public other than as a result of a disclosure by Buyer after the Closing Date in breach of its obligations hereunder, (C) are required to be disclosed by Applicable Law the rules of any selfregulatory organization so long as reasonable prior notice is given to Sellers of such disclosure and a reasonable opportunity is afforded to Seller to contest the same, (D) are disclosed in connection with the enforcement of any right or remedy relating to this Agreement or (E) are obtained independently from a source that is authorized to disclosure such data, information and documents.
- Maiver. In connection with the Contemplated Transactions, each Seller waives any and all rights of first offer, all transfer restrictions or any similar rights or restrictions set forth in the Organizational Documents of the Company, including the right of first offer set forth in Section 5.1(b) of the Limited Liability Company Operating Agreement of the Company, dated as of March 27, 2015, and any notice and/or consent requirements in connection therewith, in each case, with respect to the offer and sale of the Membership Interests pursuant to this Agreement. This waiver shall be a valid, binding and enforceable legal obligation of each Seller.

# ARTICLE VII INDEMNITY

7.1 Survival. All representations and warranties of the parties hereto contained in this Agreement or in any certificate or other document delivered pursuant to the terms of this Agreement shall survive the Closing Date until eighteen (18) months after the Closing Date, other than (a) Sections 4.10 (Company-Owned Equity Securities) and 4.12(c) (Unlawful Payments), which shall survive for three (3) years after the Closing Date; (b) Sections 4.5(b) (Undisclosed Liabilities), 4.9 (Company Loans), and 4.11 (Company Assets; Title to Company Assets), which shall survive for three (3) years after the Closing Date; (c) Sections 3.1 (Corporate Organization of Sellers), 3.2 (Authority; No Violation), 3.4 (Title to Membership Interests), 4.1 (Corporate Organization of the Company; Subsidiaries), 4.2 (Capitalization), 4.3 (Authority; No Violation), 4.6 (Broker's Fees), 4.14 (Taxes), 4.15 (No Change of Control Payments), 5.1 (Corporate Organization), 5.2 (Authority; No Violation), and 5.4 (Broker's Fees), which shall survive until sixty (60) days following the expiration of all statutes of limitations applicable to the matters covered thereby, including any extensions thereof; and (d) Section 3.7 (Representations of GALIC), which shall survive for six (6) months after the Closing Date (such representations and warranties set forth in the foregoing clauses (b), (c) and (d) are hereafter collectively referred to as the "Excepted Matters"); provided, however, that representations or warranties which are the basis for claims timely asserted under this Agreement prior to the expiration of such applicable time periods shall also survive until the final resolution of those claims. All covenants and other agreements of the parties hereto contained in this Agreement shall survive the Closing indefinitely, unless a covenant, by its terms, ends on a specified date.

#### 7.2 Indemnification.

- (a) Subject to the provisions of this Article VII, Medley shall indemnify and hold harmless Buyer, the Acquired Companies, and their respective shareholders, Subsidiaries, Affiliates and Representatives (together, the "Buyer Indemnified Parties") from, against, and in respect of any and all direct and indirect damages, losses, Liabilities, Proceedings, judgments, settlements, Liens, Taxes, penalties, interest obligations and expenses (including costs of investigation and defense and reasonable attorney and other professional advisor and consulting fees and expenses) whether or not involving a third party (collectively, "Losses") incurred or suffered by such Buyer Indemnified Party arising from, by reason of or, in connection with:
- (i) any misrepresentation or breach or alleged misrepresentation or breach of any representation or warranty of Medley contained in <u>Article III</u> or any other certificate or document delivered by Medley pursuant to this Agreement or any other Transaction Document; and
- (ii) any breach or non-performance by Medley of any covenant or agreement made by Medley in this Agreement, any other Transaction Document or any other certificate or document delivered in connection with this Agreement or any other Transaction Document.
- (b) Subject to the provisions of this Article VII, Medley, severally, shall indemnify and hold harmless the Buyer Indemnified Parties from, against and in respect of its Pro Rata Portion of any and all Losses incurred or suffered by any such Buyer Indemnified Party arising from, by reason of or, in connection with:
- (i) any misrepresentation or breach or alleged misrepresentation or breach of any representation or warranty of the Sellers contained in <u>Article IV</u>, any other Transaction Document or any other certificate or document delivered by the Company pursuant to this Agreement or any other Transaction Document (including, for the avoidance of doubt, the Purchase Price Certificate);
- (ii) any breach or non-performance by the Company of any covenant or agreement made by the Company in this Agreement, any other Transaction Document or any other certificate or document delivered in connection with this Agreement or any other Transaction Document;
  - (iii) any Pre-Closing Taxes; and
- (iv) any Indebtedness of the Acquired Companies not included in the calculation of the Purchase Price set forth in the Purchase Price Certificate.
- (c) Subject to the provisions of this <u>Article VII</u>, Buyer shall indemnify and hold harmless each Seller and its Representatives (together, the "Seller Indemnified Parties")

from, against and in respect of any and all Losses incurred or suffered by any such Seller Indemnified Party arising from, by reason of or, in connection with:

- (i) any misrepresentation or breach or alleged misrepresentation or breach of any representation or warranty of Buyer contained in <u>Article V</u>, any other Transaction Document to which Buyer is a party or any other certificate or document delivered by Buyer pursuant to this Agreement or any other Transaction Document; and
- (ii) any breach or non-performance by Buyer of any of its covenants or agreements contained in this Agreement, any other Transaction Document to which Buyer is a party or any other certificate or document delivered by Buyer pursuant to this Agreement or any other Transaction Document.
- (d) Subject to the provisions of this <u>Article VII</u>, GALIC shall indemnify and hold harmless the Buyer Indemnified Parties from, against and in respect of any and all Losses incurred or suffered by any such Buyer Indemnified Party arising from, by reason of or, in connection with any misrepresentation or breach or alleged misrepresentation or breach of any representation or warranty of GALIC contained in Section 3.7.

#### 7.3 Defense of Claims.

Non-Third Party Claims. If a claim for Losses (a "Claim") is to be made by an Indemnified Party, other than in connection with a Tax Claim (which shall be governed by Section 2.3(e)), such Indemnified Party shall give notice (including a reasonably detailed description of the Claim) to the party from whom such indemnification is sought (the "Indemnifying Party") before the applicable survival period for such Claim expires (if ever) in accordance with Section 7.1 and as promptly as practicable after such Indemnified Party becomes aware of any fact, condition or event which may give rise to Losses for which indemnification may be sought under this Article VII; provided, however, that the failure of any Indemnified Party to give timely notice hereunder shall not affect rights to indemnification hereunder, except to the extent the Indemnifying Party is actually prejudiced by such failure. Within 30 days after delivery of a notice pursuant to this Section 7.3(a) (the "Response Period"), the Indemnifying Party shall deliver to the Indemnified Party a written response to such notice. If the Indemnifying Party fails to so respond within such period, then the Indemnifying Party shall be deemed to have irrevocably accepted the notice and agreed to pay the Losses at issue therein, subject to the limitations set forth in this Article VII, and the Indemnifying Party shall satisfy such obligation promptly, and in any event within ten (10) days after the expiration of the Response Period. If, within 30 days after delivery of the notice pursuant to this Section 7.3(a), the Indemnifying Party delivers a written notice disputing the Indemnified Party's entitlement to indemnification for the Losses described in such notice, then the parties shall use their commercially reasonable efforts to settle such disputed matters within 30 days following the expiration of the Response Period. The parties hereto acknowledge and agree that the Federal Rules of Evidence Rule 408 shall apply to the parties hereto during any such negotiations and any subsequent dispute arising therefrom. If the parties are unable to reach agreement within such 30-day period, the dispute may be resolved by any legally available means consistent with the provisions of Section 9.6.

- Third Party Claims. With respect to all third party Proceedings (or threatened Proceedings) that give rise to a Claim hereunder, other than in connection with a Tax Claim (which shall be governed by Section 2.3(e)), the Indemnified Party shall promptly after obtaining actual knowledge of the assertion of any Proceeding (or threatened Proceeding) give written notice to the Indemnifying Party. After receipt of any such notice of such a Proceeding (or threatened Proceeding), the Indemnifying Party shall have the right to defend the Indemnified Party against the Proceeding with counsel of its choice reasonably satisfactory to the Indemnified Party, unless the nature of the Claim creates an ethical conflict for the same counsel to represent the Indemnified Party and the Indemnifying Party, so long as (i) the Indemnifying Party notifies the Indemnified Party in writing within thirty (30) days after the Indemnified Party has given notice of the Claim that the Indemnifying Party will indemnify the Indemnified Party from and against the entirety of any Losses the Indemnified Party may suffer resulting from, arising out of, relating to, in the nature of, or caused by the Claim or raised in the Proceeding, (ii) the Proceeding involves only a claim for money damages and no other relief, (iii) the Indemnifying Party conducts the defense of the Proceeding actively and diligently, and (iv) the Proceeding does not relate to or otherwise arise in connection with Taxes or any criminal or regulatory enforcement action. The Indemnifying Party shall not compromise or settle such Proceeding without the written consent of the Indemnified Party. The Indemnified Party shall not settle or compromise any Claim that is the subject of indemnification under this Agreement, without the prior written consent of the Indemnifying Party (such consent not to be unreasonably conditioned, withheld or delayed) if (A) injunctive or other equitable relief will be imposed against the Indemnifying Party, (B) such settlement does not expressly and unconditionally release the Indemnifying Party from all Liabilities with respect to such claim, without prejudice, or (C) there is a finding or admission of any violation of Law or any admission of Liability by any Indemnifying Party. In all other cases the Indemnified Party may defend the Claim or Proceeding with counsel of its choosing at the reasonable expense of the Indemnifying Party. The Indemnifying Party shall promptly and frequently apprise the Indemnified Party of the status of the Proceeding and shall furnish the Indemnified Party with such documents and information filed or delivered in connection with such Proceeding as the Indemnified Party may reasonably request. The Indemnified Party may, at its own cost, participate in the investigation, trial and defense of any such Proceeding defended by the Indemnifying Party and any appeal arising therefrom. The parties shall cooperate with each other in connection with any defense and in any notifications to insurers. If the Indemnifying Party fails to promptly and diligently assume and thereafter prosecute the defense of such Proceeding after receipt of notice hereunder, the Indemnified Party against which such Claim has been asserted shall (upon delivering notice to such effect to the Indemnifying Party) have the right to undertake the defense, compromise or settlement of such Proceeding with counsel of its own choosing at the reasonable expense of the Indemnifying Party and the Indemnifying Party shall have the right to participate therein at its own cost.
  - 7.4 Adjustment. The parties hereto agree that any indemnification payments made pursuant to this Agreement shall be treated for all Tax purposes as an adjustment to the Purchase Price, unless otherwise required by Applicable Law.

#### 7.5 Limitations.

- (a) Except for Losses in respect of the Excepted Matters or resulting from fraud, willful misconduct, or intentional misrepresentation, no Seller shall be liable to any Buyer Indemnified Party for Losses under Section 7.2(a)(i) or Section 7.2(b)(i), (i) unless the aggregate amount of the Losses suffered by Buyer Indemnified Parties exceeds on a cumulative basis an amount equal to \$750,000 (the "Threshold"), at which time the Buyer Indemnified Parties shall be entitled to be indemnified against the aggregate amount of Losses, regardless of the Threshold, and (ii) for an aggregate amount of Losses in excess of such Seller's proportional interest in the Purchase Price paid by Buyer (based on such Seller's Pro Rata Portion).
- (b) Except for Losses in respect of the Excepted Matters or resulting from fraud, willful misconduct, or intentional misrepresentation, Buyer shall not be liable to any Seller Indemnified Party for Losses under Section 7.2(c)(i), (i) unless the aggregate amount of the Losses suffered by the Seller Indemnified Parties exceeds on a cumulative basis the Threshold, at which time such Seller Indemnified Parties shall be entitled to be indemnified against the aggregate amount of Losses, regardless of the Threshold, (ii) with respect to Medley and its Representatives collectively, for an aggregate amount of Losses in excess of Medley's proportional interest in the Purchase Price paid by Buyer (based on Medley's Pro Rata Portion), and (iii) with respect to GALIC and its Representatives collectively, for an aggregate amount of Losses in excess of GALIC's proportional interest in the Purchase Price paid by Buyer (based on GALIC's Pro Rata Portion).
- (c) For purposes of this Article VII, a breach of a representation or warranty shall be deemed to exist either if such representation or warranty is actually inaccurate or breached or would have been inaccurate or breached if such representation or warranty had not contained any limitation or qualification as to "materiality", "Material Adverse Effect", or similar language, and the amount of Losses in respect of any breach of representation or warranty, including any deemed breach pursuant to this clause, shall be determined without regard and without giving effect to any such limitation or qualification as to "materiality", "Material Adverse Effect", or similar language set forth in such representation or warranty.
- (d) The parties shall cooperate with each other to resolve any Claim with respect to which one party is obligated to indemnify the other party hereunder, including by making commercially reasonable efforts to mitigate or resolve any such Claim.
- (e) No Indemnifying Party shall be liable pursuant to this Agreement for any punitive or exemplary damages except to the extent awarded in connection with a third party Claim.
- (f) The amount of any Losses for which indemnification is provided under this Agreement shall be net of (A) insurance proceeds actually recovered by the Indemnified Party in respect of such Losses (net of any out-of-pocket costs incurred in connection with such recovery and any increases in premium) and (B) recoveries from third parties pursuant to indemnification or otherwise (net of any out-of-pocket costs incurred in connection with such recovery.
- (g) With respect to all matters other than the Syndicated Company Loans, the right to indemnification, payment of damages and other remedies based on representations,

warranties, covenants and obligations contained in this Agreement shall not be affected by any investigation conducted or any knowledge acquired (or capable of being acquired) at any time, whether before or after the Closing Date, with respect to the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or obligation. With respect to all matters relating to the Syndicated Company Loans, to the extent Buyer or any of its Affiliates has knowledge, after due inquiry and investigation, prior to the Closing, of a fact, circumstance or event related to such Company Loan that is a Syndicated Company Loan or to a Borrower under a Syndicated Company Loan, such fact, circumstance or event shall be deemed disclosed by Sellers to Buyer for all purposes hereunder and shall not be the basis for any claim to indemnification, payment of damages or any other remedy under this Agreement. The parties agree that due inquiry and investigation of Buyer and its Affiliates shall not require inquiry of any Person that is not an Affiliate of Buyer or an officer, director, employee or representative of Buyer or an Affiliate of Buyer, does not extend beyond any such individual's areas of responsibility or such individual's direct reports (both senior and junior), and does not require inquiry or investigation beyond the books and records of Buyer or any Affiliate of Buyer other than information made available to such Person by way of a virtual data room with respect to the Syndicated Company Loans as of the Business Day immediately prior to the Closing.

7.6 Waiver. The remedies provided for in this Article VII shall be exclusive and shall preclude assertion by any Indemnified Party of any other rights or the seeking of any and all other remedies against the Indemnifying Party for claims based on this Agreement (regardless of the theory, whether in contract, at law, in equity or otherwise), other than for (a) actions for specific performance or other equitable remedies (but not rescission) referred to in Section 9.10 or (b) Losses arising out of or relating to fraud, willful misconduct or intentional misrepresentation. Each party hereby waives any provision of Applicable Law to the extent that it would limit or restrict the agreement contained in this Section 7.6. For the avoidance of doubt, this Section 7.6 shall not limit the ability of any party from making any claim arising out of or relating to fraud, willful misconduct or intentional misrepresentation.

#### ARTICLE VIII DEFINED TERMS

8.1 For purposes of this Agreement, the following terms shall have the meanings set forth below:

"Acquired Companies" means the Company, MCC Holdings and MCC SPV.

"Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under common control with such Person. The term "control" means possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise; provided, however, that in no event shall a portfolio company of an Acquired Company, whether an Existing Portfolio Company or otherwise, be deemed to be an Affiliate of an Acquired Company.

"Affiliated Group" means an affiliated group as defined in Section 1504 of the Code (or any analogous combined, consolidated or unitary group defined under state, local or non-U.S. Law relating to income Tax) of which an Acquired Company is or has been a member.

"Agreement" has the meaning set forth in the preamble to this Agreement.

"Allocation Schedule" has the meaning set forth in Section 2.3(a)(i).

"Anti-Corruption Laws" has the meaning set forth in Section 4.12(c).

"Applicable Law" means, with respect to a specified Person, any federal, state, local, municipal, or foreign constitution, treaty, law (including the common law), statute, code, ordinance, rule, regulation, permit, approval, judgment, Order, writ, decree, injunction, resolutions or requirements (collectively, "Laws") applicable to the specified Person.

"Borrowers" means those Persons who constitute "borrowers" (or any similarly defined entity) under the Company Loan Documents.

"Business Day" means a day, other than Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by Applicable Law to close.

"Buyer" has the meaning set forth in the preamble to this Agreement.

"Buyer Indemnified Parties" has the meaning set forth in Section 7.2(a).

"Buyer Prepared Tax Return" has the meaning set forth in Section 2.3(b).

"Cash and Cash Equivalents" means, without double counting, the aggregate amount of all cash and cash equivalents of the Acquired Companies as of the Closing Date.

"Claim" has the meaning set forth in Section 7.3(a).

"Closing" has the meaning set forth in Section 2.1.

"Closing Date" has the meaning set forth in Section 2.1.

"Code" means the United States Internal Revenue Code of 1986, as amended.

"Company Loan Documents" means the loan agreements, credit and financing agreements, indentures, guarantees, subordination agreements, Company Loan Notes, note purchase agreements, mortgages, deeds of trust, security agreements (including pledge and control agreements), financing statements, intercreditor agreements, mortgage, sale and servicing agreements, acquisition agreements, intercreditor agreements and other instruments, insurance policies, assumption or substitution agreements, side letters and documents affecting the Acquired Companies' ownership, economic or other rights with respect to the Company Loans or in which an Acquired Company has an interest, in connection with the Company Loans, in each case, together with all amendments and modifications thereto and any reports or documents

delivered to an Acquired Company in its capacity as a lender or noteholder under any such agreement.

"Company Loan Notes" means the original executed promissory notes (or copies, to the extent that only copies of such promissory notes are in an Acquired Company's possession or control) issued to the order of an Acquired Company, or copies of a "master" note if no such note was issued to an Acquired Company or an allonge endorsing a note in favor of an Acquired Company, evidencing indebtedness owing to an Acquired Company under a Company Loan.

"Company Loans" means the loans owned by the Acquired Companies as of the Closing Date.

"Company-Owned Equity Interests" means the Equity Interests identified on Section 4.10(a) of the Seller Disclosure Schedule as owned by the Acquired Companies.

"Confidentiality Agreement" means that certain Confidentiality Agreement, dated as of June 3, 2020, by and between Medley and Golub Capital LLC.

"Consent" means any consent of the Borrower, the administrative agent, the issuer, any co-investor or other Person required to sell, assign, transfer, convey or deliver the Membership Interests.

"Contemplated Transactions" means the transactions contemplated by this Agreement.

"Contract" means any contract, lease, sublease, license, permit, purchase and sale order, instrument, note and any other agreement, commitment or binding arrangement or understanding, whether written or oral.

"DB Credit Facility" means the credit facility provided to the Acquired Companies under the DB Loan Documents.

"DB Loan Documents" means, collectively, that certain Loan and Servicing Agreement, dated as of August 4, 2015, by and among MCC SPV, as Borrower, the Company, as Servicer and as Originator, each Lender from time to time party thereto, each Lender Agent from time to time party thereto, U.S. Bank National Association, as the Collateral Agent, Account Bank and Collateral Custodian, and Credit Suisse AG, Cayman Islands Branch, as the Administrative Agent, and each document, instrument or agreement entered into in connection with the foregoing, in each case, as amended, amended and restated, supplemented or otherwise modified from time to time.

"Enforceability Exception" has the meaning set forth in Section 3.2(a).

"Equity Governing Documents" means, with respect to a Company-Owned Equity Interest, the Organizational Document, option or warrant agreement, registration rights agreement, buy-sell arrangement and any other document that governs or otherwise affects the terms of any Company-Owned Equity Interest.

"Equity Interest" means (a) any partnership interests, (b) any membership interests or units, (c) any units of capital stock, (d) any other interest or participation that confers on a Person the right to receive a unit of the profits and losses of, or distribution of assets of, the issuing entity, including any stock appreciation, phantom stock, profit participation or similar rights, (e) any subscriptions, calls, warrants, options, or commitments of any kind or character relating to, or entitling any Person or entity to purchase or otherwise acquire membership interests or units, capital stock, or any other equity securities, (f) any securities convertible into or exercisable or exchangeable for partnership interests, membership interests or units, capital stock, or any other equity securities, or (g) any other interest classified as an equity security of a Person.

"Excepted Matters" has the meaning set forth in Section 7.1.

"Ex-Im Laws" means all U.S. and non-U.S. Laws relating to export, reexport, transfer, and import controls, including, without limitation, the Export Administration Regulations, the customs and import Laws administered by U.S. Customs and Border Protection, the EU Dual Use Regulation (428/2009), all equivalent or similar legislation to the foregoing implemented by EU Member States, and the UK's Export Control Order 2008.

"Existing Portfolio Company" means any portfolio company of the Acquired Companies listed on Sections 4.9(a) and 4.10(a) of the Seller Disclosure Schedule.

"Financial Statements" has the meaning set forth in Section 4.5(a).

"GAAP" means United States generally accepted accounting principles, consistently applied.

"GALIC" has the meaning set forth in the preamble to this Agreement.

"GALIC Interests" has the meaning set forth in the recitals to this Agreement.

"Governmental Entity" means any federal, state or local government, nation, state, province, territory, district or any court, administrative or regulatory agency or commission, arbitrator, mediator, tribunal or other governmental or quasi-governmental authority or agency, department, bureau, office, commission, organization, official or authority domestic or foreign.

"Indebtedness" means with respect to a Person, without duplication, all Liabilities, contingent or otherwise, in respect of: (a) borrowed money or issued in exchange or substitution for borrowed money (including any prepayment costs and penalties associated with prepayment of such Liabilities), (b) any indebtedness evidenced by any note, bond, debenture or other debt security, (c) all amounts due and payable under or obligations in respect of letters of credit, bankers' acceptances, bank overdrafts, surety and performance bonds or similar facilities and financial guarantees issued for the account of such Person, (d) capitalized lease obligations, (e) the deferred purchase price of assets, services or securities (other than trade accounts payable incurred in the ordinary course of business), (f) interest rate protection agreements or any interest, currency or other hedging, swap, derivative or similar hedging agreements, (g) conditional sale or other title retention agreements, (h) interest, premium, penalties and other amounts owing in respect of the items listed in the foregoing clauses (a) through (g), and (i) guarantees and surety and other similar agreements of such Person of any of the foregoing of any

other Person, including guarantees in the form of an obligation to repurchase or reimburse, whether or not any of the foregoing would appear on a consolidated balance sheet prepared in accordance with GAAP, and any fees, penalties or accrued and/or unpaid interest on the foregoing.

"Indemnified Party" means the Person or Persons entitled to, or claiming a right to, indemnification under Article VII.

"Indemnifying Party" has the meaning given to it in Section 7.3(a).

"Independent Accountant" means an independent accounting firm of national reputation.

"Interim Financial Statements" has the meaning set forth in Section 4.5(a).

"IRS" means the United States Internal Revenue Service.

"Knowledge of Medley" or "Medley's Knowledge" means the knowledge of Richard Allorto, Howard Liao, Nathan Bryce and Dean Crowe, after due inquiry and investigation, together with any knowledge (a) that such person could have been expected to discover or otherwise become aware of in the course of such due inquiry and investigation, and (b) that has been made available to any such person, including by way of a virtual data room, as of the Closing with respect to the Company Loans (other than the Syndicated Company Loans). For purposes of this definition, due inquiry and investigation of any individual listed in this definition shall not require inquiry of any Person that is not an Affiliate of Medley, Medley LLC or an officer, director, employee or representative of Medley, Medley LLC or an Affiliate of Medley or Medley LLC, does not extend beyond such individual's areas of responsibility or such individual's direct reports (both senior and junior) and does not require inquiry or investigation beyond the books and records of Medley, Medley LLC or any Affiliate of Medley LLC other than information made available by way of a virtual data room, as of the Business Day immediately prior to Closing.

"Laws" has the meaning set forth in the definition of Applicable Law.

"Liability" or "Liabilities" means a liability, Indebtedness, obligation, commitment, expense, claim or cause of action (of any kind or nature whatsoever, whether absolute, accrued, contingent or other, and whether known or unknown).

"Liens" means liens, pledges, charges, mortgages, license, option, right of first refusal, claims and security interests and similar encumbrances.

"Losses" has the meaning set forth in Section 7.2(a).

"Material Adverse Effect" means any occurrence, change, event, state of facts, effect or development that, individually, or taken together with all other occurrences, changes, events, effects or developments, (a) has or would reasonably be likely to have a material adverse effect on the condition (financial or otherwise), results of operations, properties, assets (including, without limitation, the Company Loans and Company-Owned Equity Interests) or business of the Acquired Companies taken as a whole; provided, however, that, with respect to this subsection

(a), the determination of whether a "Material Adverse Effect" exists or has occurred shall not include effects to the extent attributable to (i) changes, after the date hereof, in GAAP or regulatory accounting requirements applicable generally to companies in the industry in which the Acquired Companies operate, (ii) changes, after the date hereof, in laws, rules or regulations of general applicability to companies in the industry in which the Acquired Companies operate, (iii) changes in global or national political conditions or general economic or market conditions generally affecting other companies in the industry in which the Acquired Companies operate, (iv) conditions arising out of acts of terrorism, war, weather conditions, pandemics or other similar events, or (v) the public disclosure of this Agreement or the Contemplated Transactions or any action required to be taken in accordance with this Agreement (other than covenants to operate in the ordinary course of business), except, with respect to clauses (i), (ii), (iii) and (iv), to the extent that the effects of such change are disproportionately adverse to the condition (financial or otherwise), results of operations, properties, assets (excluding, without limitation, the Syndicated Company Loans) or business of the Acquired Companies, taken as a whole, as compared to other companies in the industry in which the Acquired Companies operate, or (b) with respect to either Seller or Buyer, has or would reasonably be expected to prevent or materially delay the ability of either Seller or Buyer, as applicable, to timely consummate the Contemplated Transactions or otherwise perform their respective obligations under the Transaction Documents.

"MCC Holdings" has the meaning set forth in Section 4.1(d).

"MCC SPV" has the meaning set forth in Section 4.1(d).

"Medley" has the meaning set forth in the preamble to this Agreement.

"Medley Interests" has the meaning set forth in the recitals to this Agreement.

"Membership Interest Assignment" has the meaning in Section 2.2(a)(i).

"NYSE" means the New York Stock Exchange.

"OFAC" has the meaning set forth in the definition of Sanctioned Person.

"Order" means any final award, injunction, judgment, order, ruling, assessment, decree, writ, determination or verdict entered or issued by any court of Governmental Entity pertaining to any action, suit, proceeding, arbitration, mediation, audit or investigation.

"Organizational Documents" means, with respect to a Person other than a natural person, (a) the articles or certificate of incorporation and the bylaws of a corporation; (b) the certificate of formation and operating agreement of a limited liability company; (c) the partnership agreement and any statement of partnership of a general partnership; (d) the limited partnership agreement and the certificate of limited partnership of a limited partnership; (e) any charter or similar document adopted or filed in connection with the creation, formation, or organization of any other Person; (f) any stockholder or similar agreement among holders of securities of an issuer; and (g) any amendment to any of the foregoing.

"Payoff Letter" has the meaning set forth in Section 2.2(a)(vii).

"Permit" means any license, permit, variance, exemption, franchise, consent, approval, authorization, qualification, or Order of any Governmental Entity.

"Person" means an individual, corporation, partnership, limited liability company, association, trust, sole proprietorship, unincorporated organization, other entity, organization, group (as defined in Section 13(d) of the Securities Exchange Act of 1934, as amended), or any other business entity or any Governmental Entity, including a government or political subdivision or an agency or instrumentality thereof.

"Post-Closing Tax Period" means any taxable period beginning after the Closing Date or, with respect to Straddle Period, the portion of such Straddle Period beginning after the Closing Date.

"Pre-Closing Tax Period" means any taxable period ending at or prior to the Closing Date or, with respect to any Straddle Period, the portion of such Straddle Period ending at the Closing Date.

"Pre-Closing Taxes" means (a) any Taxes (or the non-payment thereof) of, or with respect to, an Acquired Company for any Pre-Closing Tax Period (determined for any Straddle Period in accordance with the principles set forth in Section 2.3(c)), including any Tax assessment resulting from application of the centralized partnership audit regime enacted by the Bipartisan Budget Act of 2015 and any partnership tax withholding obligations, and (b) all Taxes of any Person for a Pre-Closing Tax Period for which an Acquired Company is liable under Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or federal Law), as a transferee or successor, under any Tax agreement or by contractual obligation (other than Contracts entered into in the ordinary course of business and not primarily related to Taxes).

"Pro Rata Portion" has the meaning set forth in Section 1.1(a).

"Proceeding" means any action, claim, arbitration, audit, hearing, investigation, litigation, suit or other proceeding (whether civil, criminal, administrative, investigative, or informal) commenced, brought, conducted, or heard by or before, or otherwise involving, any court or other Governmental Entity or referee, trustee, arbitrator or mediator.

"Purchase Price" has the meaning set forth in Section 1.1(b).

"Purchase Price Certificate" has the meaning set forth in Section 1.1(b).

"Representative" means, with respect to any Person, its Affiliates, directors, managers, officers, employees, attorneys, accountants, financing sources, financial advisors, agents, consultants or other representatives.

"Response Period" has the meaning set forth in Section 7.3(a).

"Sanctioned Country" means any country or region that is, or has been in the last five (5) years, the subject or target of a comprehensive embargo under Sanctions Laws (including Cuba, Iran, North Korea, Sudan, Syria, and the Crimea region of Ukraine).

"Sanctioned Person" means any individual or entity that is the subject or target of sanctions or restrictions under Sanctions Laws or Ex-Im Laws, including: (a) any individual or entity listed on any applicable U.S. or non-U.S. sanctions- or export-related restricted party list, including, without limitation, the U.S. Department of the Treasury Office of Foreign Assets Control's ("OFAC") Specially Designated Nationals and Blocked Persons List; the EU Consolidated List and the UK's Consolidated List of Persons Subject to Financial Sanctions; (b) any entity that is, in the aggregate, fifty percent (50%) or greater owned, directly or indirectly, or otherwise controlled by a Person or Persons described in clause (a); (c) any individual or entity acting on behalf of or at the direction of any Persons described in clauses (a) or (b); or (d) any national of a Sanctioned Country.

"Sanctions Laws" means all U.S. and non-U.S. Laws relating to economic or trade sanctions, including, without limitation, the Laws administered or enforced by the United States (including by OFAC or the U.S. Department of State), the United Nations Security Council, the European Union, the United Kingdom and all other applicable EU Member States.

"SEC" means the United States Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended.

"Seller" has the meaning set forth in the preamble to this Agreement.

"Seller Disclosure Schedule" means that certain disclosure schedule delivered by the Sellers to Buyer prior to the execution of this Agreement, which schedule sets forth, among other things, items the disclosure of which is necessary or appropriate either in response to an express disclosure requirement contained in <a href="Articles III">Articles III</a> and <a href="IV">IV</a> or as an exception to one or more representations or warranties contained in <a href="Articles III">Articles III</a> and <a href="IV">IV</a>, or to one or more of the Sellers' covenants contained in this Agreement, to the extent the relevance of such disclosure to such section or subsection is reasonably apparent on its face.

"Seller Indemnified Parties" has the meaning set forth in Section 7.2(c).

"Seller Prepared Tax Return" has the meaning set forth in Section 2.3(b).

"Solvent" means, at any time with respect to any Person, on the applicable date of determination, such Person (a) is able to pay its debts as they mature and has (and has a reasonable basis to believe it will continue to have) sufficient capital (and not unreasonably small capital) to carry on its business consistent with its practices as of the Closing Date, and (b) the assets and properties of such Person at a fair valuation (and including as assets for this purpose at a fair valuation all rights of subrogation, contribution or indemnification arising pursuant to any guarantees given by such Person) are greater than the Indebtedness of such Person, and including subordinated and contingent liabilities computed at the amount which, such Person has a reasonable basis to believe, represents an amount which can reasonably be expected to become an actual or matured liability (and including as to contingent liabilities arising pursuant to any guarantee the face amount of such liability as reduced to reflect the probability of it becoming a matured liability).

"Straddle Period" means any taxable period beginning before the Closing Date and ending after the Closing Date.

"Subsidiary" means, with respect to any Person, any corporation, company, limited liability company, partnership, association, or other business entity of which (a) if a corporation or a company, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers, or trustees thereof is at the time owned or controlled, directly or indirectly, by such Person or one or more of the other Subsidiaries of such Person or a combination thereof or (b) if a limited liability company, partnership, association, or other business entity (other than a corporation or a company), a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by such Person or one or more Subsidiaries of such Person or a combination thereof and for this purpose, a Person or Persons own a majority ownership interest in such a business entity (other than a corporation or a company) if such Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be a, or control any, managing director or general partner of such business entity (other than a corporation or a company); provided, however, the term "Subsidiary" shall not include any portfolio company of the Acquired Companies, whether an Existing Portfolio Company or otherwise.

"Syndicated Company Loans" means the Company Loans that are set forth on <u>Section</u> 8.1 of the Seller Disclosure Schedule.

"Tax" or "Taxes" means (a) all federal, state, local, and foreign income, excise, gross receipts, recording, gross income, ad valorem, profits, gains, property, stamp, capital, sales, transfer, use, payroll, employment, severance, withholding, social security (or similar including FICA), customs, duties, intangibles, franchise, backup withholding, escheat or unclaimed property, registration, value added, alternative or add-on minimum, estimated, and any other taxes, charges, levies or like assessments in the nature of or similar to taxes, together with all penalties and additions to tax and interest thereon, in each case whether disputed or not and (b) any liability for Taxes described in clause (a) above for the payment of any amounts as a result of being liable for another Person's taxes as a transferee or successor, by contract or otherwise.

"Tax Return" means, with respect to a Person, a report, return, election, declaration, or other information or statement (including any amendments thereof and attachments thereto) supplied or required to be supplied to a Governmental Entity with respect to Taxes including, where permitted or required, combined or consolidated returns for any group of entities that includes the Person or any of its Subsidiaries.

"Threshold" has the meaning set forth in Section 7.5(a).

"Trade Control Law" has the meaning set forth in Section 4.12(b).

"Transaction Documents" means this Agreement, the Membership Interest Assignment Agreements and any other document or certificate executed and delivered in connection with the foregoing.

"Transfer Taxes" means, collectively, all federal, state, local foreign transfer, documentary, excise, sales, use, value added, registration, stamp, recording, property and similar Taxes or fees, including any penalties and interest.

#### ARTICLE IX GENERAL PROVISIONS

- 9.1 Expenses. Except as provided in Section 2.3(f) with respect to Transfer Taxes, the Sellers shall bear their (and the Company's) costs, fees and expenses, including attorney, investment banker, broker, accountant and other representative and consultant fees, incurred in connection with the execution and negotiation of the Transaction Documents and the consummation of the transactions contemplated hereby and thereby; and Buyer shall bear its own costs, fees and expenses, including attorney, investment banker, broker, accountant and other representative and consultant fees, incurred in connection with the execution and negotiation of the Transaction Documents and the consummation of the transactions contemplated hereby and thereby.
- 9.2 <u>Notices</u>. All notices, demands or other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and shall be deemed to have been given (a) when delivered personally to the recipient, (b) when sent by facsimile, electronic mail or other electronic transmission followed by personal delivery, delivery by certified mail or delivery by reputable overnight courier service, on the date sent by facsimile if sent during normal business hours and the next Business Day if sent after normal business hours, or (c) one Business Day after being sent to the recipient by reputable overnight courier service (charges prepaid). Such notices, demands and other communications shall be sent to Buyer, the Company or the Sellers at the addresses indicated below or to such other address or to the attention of such other Person as the recipient party has specified by prior written notice to the sending party:
  - (a) if to the Sellers, to:

Medley Capital Corporation 375 Park Avenue, 33rd Floor New York, NY 10152 Attention: Richard Allorto e-mail: richard.allorto@mdly.com

with a copy to:

Kramer Levin Naftalis & Frankel LLP 1177 Avenue of the Americas New York, NY 10036 Attention: George Silfen; David S. Berg e-mail: gsilfen@kramerlevin.com; dberg@kramerlevin.com Great American Life Insurance Company 301 East Fourth Street, 27th Floor Cincinnati, Ohio 45202 Attention: Chester M. Eng and John S. Fronduti e-mail: ceng@amfin.com; jfronduti@afmin.com

(b) if to Buyer or the Company, to:

Golub Capital LLC
200 Park Avenue
New York, NY 10166
Attention: Spyro Alexopoulos; Benjamin Guilford
e-mail: salexopoulos@golubcapital.com;
bguilford@golubcapital.com

with a copy to:

Proskauer Rose LLP Eleven Times Square New York, NY 10036 Attention: Justin Breen; William J. Tuttle

e-mail: JBreen@proskauer.com; WTuttle@proskauer.com

9.3 Interpretation. When a reference is made in this Agreement to Articles, Sections, Exhibits or Schedules, such reference shall be to an Article or Section of or Exhibit or Schedule to this Agreement unless otherwise indicated. References to documents, instruments or agreements shall be deemed to refer as well to all addenda, exhibits and schedules thereto. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The Seller Disclosure Schedule, as well as all other schedules and all exhibits hereto, shall be deemed part of this Agreement and included in any reference to this Agreement. This Agreement shall not be interpreted or construed to require any person to take any action, or fail to take any action, if to do so would violate any Applicable Law. When used, the singular number includes the plural number and vice versa. Reference to any agreement, document or instrument means such agreement, document or instrument as amended, modified, supplemented, restated or replaced and in effect from time to time in accordance with the terms thereof and reference to any promissory note includes any promissory note that is an extension or renewal thereof or a substitute or replacement therefor. Reference to any law means such law as amended, modified, codified, replaced or reenacted, in whole or in part, and in effect from time to time, including rules and regulations promulgated thereunder, and reference to any section or other provision of any law means that provision of such law from time to time in effect and constituting the substantive amendment, modification, codification, replacement or reenactment of such section or other provisions. "Hereunder," "hereof," "hereto," and words of similar import shall be deemed references to this Agreement as a whole and not to any particular section or other provision hereof. The words "asset" and "property"

shall be construed as having the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.

- 9.4 Counterparts; Delivery by Facsimile or PDF. This Agreement may be executed in one or more counterparts (including by means of telecopied signature pages or signature pages delivered by electronic transmission in portable document format (pdf)), all of which taken together shall constitute one and the same instrument. This Agreement and any signed agreement or instrument entered into in connection with this Agreement, and any amendments hereto or thereto, to the extent signed and delivered by means of a facsimile machine or electronic transmission in portable document format (pdf), shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. No party hereto or to any such agreement or instrument shall raise the use of a facsimile machine or electronic transmission in portable document format (pdf) to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine or electronic transmission in portable document format (pdf) as a defense to the formation of a contract and each such party forever waives any such defense, except to the extent such defense related to lack of authenticity.
- 9.5 Entire Agreement. This Agreement (including the documents and the instruments referred to in this Agreement), together with the Confidentiality Agreement, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter of this Agreement, other than the Confidentiality Agreement.

#### 9.6 Governing Law; Jurisdiction.

- (a) This Agreement and all claims or causes of action (whether in contract, tort or statute) that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance of this Agreement (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement as an inducement to enter this Agreement) shall be governed and construed in accordance with the internal laws of the State of Delaware.
- (b) The parties hereto agree that any Proceeding brought by either party to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the Contemplated Transactions shall be brought in any federal or state court located in the County of New York, State of New York. Each of the parties hereto submits to the jurisdiction of any such court in any Proceeding seeking to enforce any provision of, or based on any matter arising out of, or in connection with, this Agreement or the Contemplated Transactions and hereby irrevocably waives the benefit of jurisdiction derived from present or future domicile or otherwise in such Proceeding. Each party hereto irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such Proceeding in any such court or that any such Proceeding brought in any such court has been brought in an inconvenient forum.

- 9.7 Waiver of Jury Trial. Each party acknowledges and agrees that any controversy which may arise under this Agreement is likely to involve complicated and difficult issues, and therefore each such party hereby irrevocably and unconditionally waives any right such party may have to a trial by jury in respect of any litigation directly or indirectly arising out of or relating to this Agreement or the Contemplated Transactions. Each party certifies and acknowledges that (a) no Representative of any other party has represented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing waiver, (b) each party understands and has considered the implications of this waiver, (c) each party makes this waiver voluntarily and (d) each party has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications in this Section 9.7.
- 9.8 <u>Publicity</u>. Neither the Sellers nor Buyer shall, and neither the Sellers nor Buyer shall permit any of their respective Subsidiaries to, issue or cause the publication of any press release or other public announcement with respect to, or otherwise make any public statement concerning, the Contemplated Transactions or otherwise disclose the terms of the Contemplated Transaction, without the prior consent (which consent shall not be unreasonably withheld) of Buyer, in the case of a proposed announcement or statement by either Seller, or Medley, in the case of a proposed announcement or statement by Buyer; <u>provided</u>, <u>however</u>, that Medley may, without the prior consent of Buyer (but after prior consultation with Buyer and opportunity to reasonably comment) issue or cause the publication of any press release or other public announcement to the extent required by Applicable Law or by the rules and regulations of the NYSE.
- 9.9 Assignment; Third Party Beneficiaries. Neither this Agreement nor any of the rights, interests or obligations under this Agreement shall be assigned by either of the parties (whether by operation of law or otherwise) without the prior written consent of the other party. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by each of the parties and their respective successors and assigns. This Agreement (including the documents and instruments referred to in this Agreement) is not intended to and does not confer upon any person other than the parties hereto any rights or remedies under this Agreement except, with respect to Article VII, the Indemnified Parties. Buyer and the Sellers hereby agree that their respective representations, warranties and covenants set forth herein are solely for the benefit of the other parties hereto, in accordance with and subject to the terms of this Agreement, and this Agreement is not intended to, and does not, confer upon any person other than the parties hereto any rights or remedies hereunder, including, without limitation, the right to rely upon such representations and warranties set forth herein. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of the parties hereto.

#### 9.10 Remedies.

(a) Except as otherwise provided in this Agreement, any and all remedies herein expressly conferred upon a party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy.

- (b) The parties hereto agree that if any provision of this Agreement were not performed in accordance with the terms hereof or is otherwise breached, irreparable damage would occur, money damages would not provide an adequate remedy at law and damages would be difficult to determine. Accordingly, it is agreed that the party or parties not in breach shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof in any court specified in Section 9.6, in addition to any other remedy to which they are entitled at law or in equity. The parties hereto hereby waive, in any such action for specific performance, the defense of adequacy of a remedy at law, the posting of any bond or other security in connection therewith and the necessity of proving actual harm.
- 9.11 <u>Amendments; Waivers</u>. No modification, amendment or waiver of any provision of, or consent required by, this Agreement shall be effective unless it is in writing and signed by the parties hereto. Any modification, amendment, waiver or consent shall be effective only in the specific instance and for the purpose for which it is given.
- 9.12 <u>Severability</u>. In the event that any provision of this Agreement as applied to any party or to any circumstance shall be adjudged by a court to be void, unenforceable or inoperative as a matter of law, then the same shall in no way affect any other provision in this Agreement, the application of such provision in any other circumstance or with respect to any other party, or the validity of enforceability of this Agreement as a whole.
- Schedules. The information disclosed in any numbered or lettered part of any Seller Disclosure Schedule shall be deemed to relate to and shall qualify the particular representation or warranty set forth in the corresponding numbered or lettered section or subsection and any other particular representation or warranty to the extent that (a) such information is cross-referenced in another part of the Seller Disclosure Schedules or (b) it is reasonably apparent on the face of the disclosure (without reference to any document referred to therein) that such information qualifies another representation and warranty of the applicable party in this Agreement. The inclusion of an item in a Seller Disclosure Schedule as an exception to a representation or warranty shall not be deemed an admission by any party hereto that such item represents an exception or material fact, event or circumstance or that such item constitutes a Material Adverse Effect. Without limiting the foregoing, no reference to or disclosure of a possible breach or violation of any contract or applicable law shall be construed as an admission or indication that such breach or violation exists or has actually occurred. Any fact or item that is disclosed in any Schedule in a way as to make its relevance or applicability to information called for by any other Schedule reasonably apparent shall be deemed to be disclosed in such other Schedule, notwithstanding the omission of a reference or cross-reference thereto.
- 9.14 Representation by Counsel. Each of the parties hereto acknowledges that it has been represented by independent counsel of its choice throughout all negotiations that have preceded the execution of this Agreement and that it has executed the same with consent and upon the advice of said independent counsel. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise, or rule of strict constriction applied, favoring or disfavoring any party hereto by virtue of the authorship of any of the provisions of this Agreement. Accordingly,

parties hereto.		

any rule of law or any legal decision that would require interpretation of any ambiguities in this Agreement against the party that drafted it is of no application and is hereby waived by the

IN WITNESS WHEREOF, the undersigned parties have caused this Agreement to be executed by their respective signatories thereunto duly authorized as of the date first above written.

#### SELLERS:

#### MEDLEY CAPITAL CORPORATION

By: /s/ Richard T. Allorto, Jr.
Name: Richard T. Allorto, Jr.
Title: Authorized Person

GREAT AMERICAN LIFE INSURANCE COMPANY

By: /s/ Mark F. Muething Name: Mark F. Muething Title: President

#### **COMPANY**:

#### MCC SENIOR LOAN STRATEGY JV I LLC

BY: MEDLEY CAPITAL CORPORATION, in its capacity as a Member of the Company

By: /s/ Richard T. Allorto, Jr.
Name: Richard T. Allorto, Jr.
Title: Authorized Person

BY: GREAT AMERICAN LIFE INSURANCE COMPANY, in its capacity as a Member of the Company

By: /s/ Mark F. Muething Name: Mark F. Muething Title: President

#### BUYER:

GEMS FUND 5, L.P.

By: GC Advisors LLC, its Investment Manager

By: /s/ Joshua M. Levinson

Name: Joshua M. Levinson

Title: Co-General Counsel and Chief

Compliance Officer

## APPENDIX A

## **Membership Interests**

Name of Seller	Membership Interests	Allocation of Purchase Price Payable to Sellers	Pro Rata Portion of Purchase Price Payable to Sellers
MEDLEY CAPITAL CORPORATION	87.5%	\$41,018,500.13	86.14 %
GREAT AMERICAN LIFE INSURANCE COMPANY	12.5%	\$ 6,601,265.33	13.86%
Total:	100%	\$ 47,619,765.46	100%

### Exhibit A

# Purchase Price Calculation

[See Attached.]

	Memo	Adj.	Final
Initial Gross Purchase Price			\$155,400,000.00
Principal Payments			
Plus: Principal Fundings since Aug 13th through Oct. 2 Less: Principal Payoffs since Aug 13th through Oct. 2 Less: Principal Payoffs since Oct. 2 through Oct. 5	_	\$0.00 (10,005,525,67) (8,750.00)	
Plus: Net Principal Changes		= -	(10,014,275.67)
Jul 1st - Sep 14th		,	8: 5:
Interest Income Accrued Interest Income Cash Received Cash Adj. for Receipts Outside of Period	\$2,382,493.71	(1,351,246.22) 684,688.41	
Interest Expense Accrued Cash Adj. for Interest Expense Paid for Jul 1st - Aug 31	(873,370.08)	729,001.48	
Net Interest Income Purchase Price Adj.	\$1,509,123.63	,	62,443.67
Sep 15th - Oct 2nd Interest Income Accrued through Oct. 2 Interest Income Cash Received through Oct. 2 Cash Adj. for Receipts Outside of Period through Oct. 2	\$560,533.96	560,533.96 (1,620,729.22) 17,162.70	
Interest Expense Accrued through Oct. 2	(182,129.95)	(182,129.95)	
Net Interest Income Purchase Price Adj.	\$378,404.01		(1,225,162.51)
Oct 3rd - Closing Est. Net Investment Income Adjustment (\$18,663	Per Day)		93,315.00
Working Capital			7-
Plus: Other Current Assets at Oct. 7 Less: Other Current Liabilities Oct. 7	_	1,658.05 0.00	
Plus: Net Working Capital			1,658.05
Plus: Purchase Price Adjustment			1,000,000.00
Adjusted Gross Purchase Price		,	\$145,317,978.54
Less: Deutsche Bank Payoff Amount (as of Oct. 7)			(111,319,482.48)
Deutsche Bank AG, New York Branch Fees Cadwalader, Wickersham & Taft LLP (Related to DB Line) US Bank Interest Calculation	_	(376,605.93) (15,000.00) (5,075.63)	et
Less: Other Deutsche Bank Payoff Fees (as of Oct. 7)			(396,681.56)
Adj. Net Purchase Price			\$33,601,814.50
Plus: Gross Cash Balance at Oct. 2 Plus: Other Operating Cash Changes from Oct. 2 through Oct. 5 $$			14,213,435.78 (195,484.82)
Total Net Proceeds to Existing Shareholders			\$47,619,765.46

## Exhibit B

# Form of Purchase Price Certificate

[See Attached.]

This Purchase Price Certificate is made and delivered Pursuant to that certain Membership Interest Purchase Agreement dated as of October 8, 2020 (the "Agreement"), by and among Medley Capital Corporation, a Delaware corporation ("Medley"), Great American Life Insurance Company, an Ohio corporation ("GALIC") (Medley and GALIC, each, a "Seller" and collectively, the "Sellers"), MCC Senior Loan Strategy JV I LLC, a Delaware limited liability company (the "Company"), and GEMS Fund 5, L.P., a Delaware limited partnership ("Buyer"). Capitalized terms used but not otherwise defined in this Purchase Price Certificate will have the same meanings ascribed to such terms in the Agreement.

The undersigned, Richard Allorto, does hereby certify to Buyer on behalf of Medley as follows:

- 1. He is the duly elected, qualified and acting Chief Financial Officer of Medley.
- 2. The calculation of the Purchase Price, attached to the Agreement as <u>Exhibit A</u>, is true and correct as of the Closing Date, in accordance with the terms of the Agreement.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the undersigned has executed this Purchase Price Certificate effective as of the Closing.

By:		
Name:	Richard Allorto	
Title:	Chief Financial Officer	

MEDLEY CAPITAL CORPORATION

## Exhibit C

# Form of Membership Interest Assignment $[S\!e\!e\,Attached.]$

#### MEMBERSHIP INTEREST ASSIGNMENT AGREEMENT

THIS MEMBERSHIP INTEREST ASSIGNMENT AGREEMENT (this "<u>Assignment</u>") dated as of October 8, 2020, is entered into by and between [•], a Delaware corporation ("<u>Assignor</u>") and GEMS Fund 5, L.P., a Delaware limited partnership ("Assignee").

WHEREAS, Assignor desires to transfer and assign, and Assignee desires to accept, [•]% of the issued and outstanding limited liability company membership interests in MCC Senior Loan Strategy JVILLC, a Delaware limited liability company (the "Company"), owned by Assignor (the "Membership Interests");

WHEREAS, after giving effect to such assignment, Assignor will own zero percent (0%) of the Membership Interests; and

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Assignor and Assignee hereby agree as follows:

- Section 1. <u>Transfer and Assignment</u>. Assignor hereby sells, transfers, assigns and conveys to Assignee, free and clear of all liens, claims, pledges and encumbrances of any nature whatsoever, all of Assignor's right, title and interest in and to the Membership Interests, to have and to hold the same unto Assignee (including Assignor's obligations under that certain Limited Liability Company Operating Agreement of the Company, dated as of March 27, 2015 (as amended, supplemented or otherwise modified from time to time, the "LLC Agreement")).
- Section 2. Acceptance. Assignee hereby accepts the foregoing assignment of the Membership Interests from Assignor, agrees to be party to and bound by the terms and conditions of the LLC Agreement and assumes all of the duties, responsibilities and rights, and agrees to perform and discharge all of the obligations of Assignor in Assignee's capacity as a member of the Company in respect of the Membership Interests conveyed. Assignor and Assignee agree that the Assignment will satisfy Section 3.1(b) of the LLC Agreement.
- Section 3. <u>Substitution</u>. Assignee is hereby admitted to the Company as a member, it being understood that Assignee is a substitute member of Assignor.
- Section 4. Further Assurances. Each party hereby agrees to do, execute, acknowledge and deliver, or to cause to be done, executed, acknowledged and delivered, all such further acts, deeds, assignments, transfers, conveyances, powers of attorney and other documents or instruments that may be reasonably requested by the other party or the Company in order to effect or confirm the assignment, acceptance and assumption provided for above.
- Section 5. Terms of the Purchase Agreement. Reference is hereby made to that certain Membership Interest Purchase Agreement dated of even date herewith by and among the Assignee, the Assignor and the other parties named therein (the "Purchase Agreement"). The terms of the Purchase Agreement, including, but not limited to, the representations, warranties, covenants and agreements relating to the Membership Interests are incorporated herein by reference. The parties hereto acknowledge and agree that the representations, warranties, covenants and agreements contained in the Purchase Agreement shall not be superseded hereby but shall remain in full force and effect to the full extent provided therein. In the event of any conflict or inconsistency between the terms of the Purchase Agreement and the terms hereof, the terms of the Purchase Agreement shall govern.

Section 6. Miscellaneous.

- (a) Governing Law. This Agreement shall be governed and construed in accordance with the laws of the State of Delaware applicable to contracts made and to be performed within such State
- (b) Headings. Section headings used in this Agreement are for convenience of reference only and shall not be used in interpreting, construing or affecting the meaning or construction of this Agreement.
- (c) Counterparts. This Agreement may be executed by the parties hereto in any number of counterparts, each of which shall be deemed to be an original but all of which together shall constitute but one and the same instrument. Any counterpart may be executed by facsimile signature and such facsimile signature shall be deemed an original.
- (d) Third Party Beneficiaries. The Company is an express third party beneficiary of this Agreement.
- (e) Binding Effect. The respective rights and obligations set forth in this Agreement shall be binding on and inure to the benefit of the parties hereto and their respective successors and assigns.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the day and year first written above.

ASSIGNOR:
[•]
By: Name: Title:
ASSIGNEE:
GEMS FUND 5, L.P., a Delaware limited partnership
By: Name:
Title:

[Signature Page to Membership Interest Assignment Agreement]

Name: Title:			

# Exhibit D

# Form of Resignation Letter [See Attached.]

#### LETTER OF RESIGNATION

Medley Capital Corporation 375 Park Avenue, 33rd Floor New York, NY 10152

October 8, 2020

To Whom It May Concern:

Reference is hereby made to that certain Membership Interest Purchase Agreement, dated as of October 8, 2020 (as amended, supplemented and/or otherwise modified from time to time, the "<u>Purchase Agreement</u>"), by and among Medley Capital Corporation, a Delaware corporation, Great American Life Insurance Company, an Ohio corporation, MCC Senior Loan Strategy JV I LLC, a Delaware limited liability company, and GEMS Fund 5, L.P., a Delaware limited partnership. Capitalized terms used, but not otherwise defined herein, shall have the meanings set forth in the Purchase Agreement.

Effective as of the Closing, the undersigned hereby irrevocably resigns from any and all director, manager and officer positions that the undersigned holds with respect to the Acquired Companies, without need for acceptance or any further action by any of the Acquired Companies.

[Signature Page Follows]

	Sincerely,
	Name: Title:
NOWLEDGED:	
SENIOR LOAN STRATEGY JV I LLC MEDLEY CAPITAL CORPORATION, in its capacity as a Member of the Company	
Name: Title:	
GREAT AMERICAN LIFE INSURANCE COMPANY, in its capacity as a Member of the Company	
Name: Title:	
	SENIOR LOAN STRATEGY JV I LLC MEDLEY CAPITAL CORPORATION, in its capacity as a Member of the Company  Name: Title: GREAT AMERICAN LIFE INSURANCE COMPANY, in its capacity as a Member of the Company  Name:

Ex		 L	T
HV	m		н
$-\alpha$	ш		-

Payoff Letter

[See Attached.]

#### TERMINATION LETTER

October 8, 2020

MCC JV SPV Funding I LLC c/o Medley Capital LLC 280 Park Avenue, 6th Floor New York, NY 10017 Attention: The General Counsel

Re: Loan and Servicing Agreement, dated as of August 4, 2015 (as amended, modified, waived, supplemented or restated from time to time, the "Loan and Servicing Agreement") by and among MCC JV SPV Funding I LLC, as the borrower (the "Borrower"), MCC Senior Loan Strategy JV I LLC, as the servicer (in such capacity, the "Servicer"), as the equityholder (in such capacity, the "Equityholder") and the originator (in such capacity, the "Originator"), Deutsche Bank AG, New York Branch, as the facility agent (in such capacity, the "Facility Agent"), each of the Lenders from time to time party thereto (the "Lender Agents"), each of the Lender Agents from time to time party thereto (the "Lender Agents") and U.S. Bank National Association, as the collateral agent (in such capacity, the "Collateral Agent"), as the account bank (in such capacity, the "Account Bank") and as the collateral custodian (in such capacity, the "Collateral Custodian").

#### Ladies and Gentlemen:

Reference is made to the Loan and Servicing Agreement described above. We have been requested to provide this letter setting forth the amount necessary, as of October 8, 2020 (the "<u>Termination Date</u>"), to pay all Obligations of the Borrower under the Loan and Servicing Agreement. Capitalized terms used but not otherwise defined herein shall have the respective meanings provided therefor in the Loan and Servicing Agreement.

1. In accordance with the foregoing, the parties hereto request that the Borrower hereby pay to the Collateral Agent the amounts set forth on Schedule I to be paid for distribution (a) to each Lender under the Loan and Servicing Agreement, an amount equal to the applicable amount set forth on Schedule I hereto in repayment of all outstanding principal, accrued and unpaid interest, fees and all other Obligations owed to the Lenders under the Loan and Servicing Agreement as of the Termination Date; (b) to counsel for the Lenders and other Secured Parties, an amount equal to the applicable amount set forth on Schedule I hereto in repayment of all outstanding fees owed to such counsel as of the Termination Date; and (c) to the Secured Parties, an amount equal to the applicable amount set forth on Schedule I hereto in repayment of all accrued and unpaid interest and fees owed to the Secured Parties as of the Termination Date (such amounts referenced in this Section 1 (and as more fully specified on Schedule I hereto) in the aggregate, the "Termination Amount"). The Collateral Agent shall disburse any amounts that are received pursuant to this Section 1 in accordance with the instructions set forth on Schedule II immediately and, in any event, no later than (x) on the same Business Day that such amounts are received if they are received at or prior to 11:00 a.m. (New York City time) on such Business

USActive 55333093.5

Day and (y) on the Business Day following receipt of such amounts if they are received after 11:00 a.m. (New York City time). The parties hereto agree that the payment of the Termination Amount shall constitute payment in full of all Obligations, except for any Obligations of the Borrower with respect to indemnification and other similar provisions in the Transaction Documents which by their express terms survive the termination of such Transaction Documents.

- 2. Each of the Borrower, the Originator, the Equityholder, the Servicer, the Collateral Agent, the Account Bank, the Collateral Custodian and the Facility Agent for itself on behalf of the Lenders hereby waives all conditions set forth in the Loan and Servicing Agreement and the other Transaction Documents in connection with the voluntary prepayment and permanent reduction of the Maximum Commitment in whole and the termination of the Loan and Servicing Agreement.
- Each Secured Party hereby acknowledges receipt of payment of all Obligations owing to it under the Transaction Documents as of the Termination Date. The parties hereto agree and acknowledge that upon receipt by the Collateral Agent of the amounts set forth in Section 1 above (and as more fully specified on Schedule I hereto), (i) all the security interests, liens and pledges in favor of the Collateral Agent or any other Secured Party securing the abovereferenced Obligations and Collateral under the Transaction Documents are automatically released with no further action on the part of any person, (ii) all obligations of the Borrower, the Originator, the Equityholder and the Servicer in respect of the Transaction Documents are deemed to be paid off, satisfied and discharged in full, (iii) the Commitments are terminated, (iv) each Transaction Document is deemed terminated and of no further force and effect, except for indemnification and other provisions thereof which by their express terms survive the termination of such Transaction Document and (v) all obligations of the Borrower, the Originator, the Equityholder and the Servicer to the Secured Parties under and/or arising in connection with the Transaction Documents are deemed to be terminated, satisfied and discharged in full and of no further force and effect, except for indemnification and other provisions thereof, which by their express terms survive the termination of the Transaction Documents.
- 4. Upon the termination of all Commitments and payment in full of the Obligations (other than unmatured contingent indemnification obligations) as described herein, the Borrower hereby requests the release of, and the Collateral Agent, the Account Bank and the Collateral Custodian shall release, all Collateral (including any Required Loan Documents and Loan Files) and other documents to the Borrower (and/or its designee). The foregoing shall be at the sole cost and expense of the Borrower with no liability to the Collateral Agent, the Account Bank or the Collateral Custodian.
- 5. The Facility Agent, the Lenders and the Secured Parties hereby acknowledge that the recipient hereof will rely on this letter and the acknowledgments, certifications, confirmations and agreements of the Secured Parties contained herein. The Secured Parties hereto authorize the filing of termination statements on form UCC-3 (including the UCC-3s attached hereto as Exhibit III) or other security interest terminations with respect to the Transaction Documents and shall procure, deliver or execute and deliver all further instruments

and documents, and take any other actions, which are reasonably required to evidence the consummation of the payoff contemplated hereby and the termination of all security interests, liens and pledges arising under or relating to the Transaction Documents, and the prompt delivery to the Borrower (or its designee) of all promissory notes, possessory collateral and other instruments in the possession of the Administrative Agent, the Collateral Agent, any Lender or any other Secured Party.

- 6. This letter may be executed by the parties hereto individually or in any combination, in one or more counterparts, each of which shall be an original and all of which shall constitute one and the same letter. Delivery of an executed counterpart of a signature page to this letter by e-mail in portable document format (.pdf) or facsimile shall be effective as delivery of a manually executed counterpart of this letter.
- 7. THE BORROWER AND EACH SECURED PARTY HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THIS LETTER OR THE TRANSACTIONS CONTEMPLATED HEREBY.
- 8. This letter shall be governed by, and construed in accordance with, the laws of the State of New York.
- The Facility Agent hereby authorizes and directs U.S. Bank National Association, in its capacity as Collateral Agent, Account Bank and as Collateral Custodian under the Loan and Servicing Agreement, to execute, deliver and perform this letter. Each of the other parties agrees that the Collateral Agent, the Account Bank and the Collateral Custodian will have no liability for entering into this letter or for any action or inaction of the Collateral Agent, the Account Bank and the Collateral Custodian, as applicable, taken pursuant to this letter (including any instruction given to the Collateral Agent, the Account Bank and the Collateral Custodian, as applicable) in accordance herewith, except to the extent that such action or inaction arises from the bad faith, gross negligence or willful misconduct of the Collateral Agent, the Account Bank and the Collateral Custodian, as applicable. Each of the other parties hereby waives and releases, to the fullest extent permitted by law, any liabilities, obligations, duties, promises or claims it may have on or prior to the Termination Date under any Transaction Document or this letter, at law or in equity, against the Collateral Agent, the Account Bank and the Collateral Custodian, as applicable, except, with respect to this letter, any claim arising from the bad faith, gross negligence or willful misconduct of the Collateral Agent, the Account Bank and the Collateral Custodian, as applicable, as described in the preceding sentence. The Collateral Agent, the Account Bank and the Collateral Custodian may conclusively rely on the directions and certifications set forth in this letter and any Transaction Document without further investigation.
- 10. The parties agree that this letter may be executed and delivered by electronic signatures and that the signatures appearing on this letter are the same as handwritten signatures for the purposes of validity, enforceability and admissibility.

3

[Signature pages to follow]

USActive 55333093.5 4

The instructions set forth above are irrevocable instructions which can only be changed by an instruction signed by the parties hereto. Please sign below in the space provided to confirm receipt and acknowledgement of foregoing.

Sincerely,

DEUTSCHE BANK AG, NEW YORK BRANCH, as Facility Agent

By:

Name: Title: Managing Director

Amit Patel

Name: Title:

Andrew Goldsmith Director

DEUTSCHE BANK AG, NEW YORK BRANCH, as Lender

Name: Amit Patel Title: Managing Director

Name:

Andrew Goldsmith Title: Director

### U.S. BANK NATIONAL ASSOCIATION, as

Collateral Agent, Account Bank and as Collateral Custodian

By:

Ralph J.

Title:

Senior Vice President

ACKNOWLEDGED AND AGREED as of the date first set forth above:

MCC JV SPV FUNDING I LLC, as Borrower

By:

Name: Richard T. Allorto, Jr. Title: Authorized Person

MCC SENIOR LOAN STRATEGY JV I LLC, as Equityholder

By:

Name: Richard T. Allorto, Jr. Title: Authorized Person

MCC SENIOR LOAN STRATEGY JV I LLC, as Servicer

Bv:

Name: Richard T. Allorto, Jr. Title: Authorized Person

MCC SENIOR LOAN STRATEGY JV I LLC, as Originator

By:

Name: Richard T. Allorto, Jr. Title: Authorized Person

# MEDLEY CAPITAL CORPORATION, as Retention Holder

By:

Name: Richard T. Allorto, Jr. Title: Chief Financial Officer

#### SCHEDULE I

## (Amounts Determined as of the Termination Date)

### 1. Outstanding Principal:

Recipient	Amount
Deutsche Bank AG, New York Branch, as Lender	\$111,319,482.48
Total	\$111.319.482.48

### 2. Accrued and Unpaid Interest and Fees:

Recipient	Amount
Deutsche Bank AG, New York Branch	\$376,605.93
Cadwalader, Wickersham & Taft LLP	\$15,000.00
U.S. Bank National Association	\$5,075.63

### 3. Total Termination Amount:

	<u>Amount</u>
-	\$111,716,164.04

#### SCHEDULE III

(See attached.)

[Schedule III]

A. NAN	ME & PHONE OF CONTACT AT FILER (optional)					
B. E-M.	IAIL CONTACT AT FILER (optional)		_			
C. SEN	ND ACKNOWLEDGMENT TO: (Name and Address)		_			
Γ		_	٦			
L	6	_	_   	- and an in F		- NII V
	IAL FINANCING STATEMENT FILE NUMBER		— quite	STATEMENT AME	R FILING OFFICE USE ( ENDMENT is to be filed [for RECORDS	
<b>/</b> T	53388054 filed on 08/05/2015 ERMINATION: Effectiveness of the Financing Statement	identified above is termina	Filer: attach Amendn	nent Addendum (Fo	rm UCC3Ad) and provide Debto	
-	Statement  ASSIGNMENT (full or partial): Provide name of Assignee	in item 7a or 7h, and addre	see of Assignee in Itam 7c and	name of Assignor	in item 0	
F	or partial assignment, complete items 7 and 9 and also ind	icate affected collateral in i	tem 8			
	CONTINUATION: Effectiveness of the Financing Stateme ontinued for the additional period provided by applicable lar		spect to the security interest(s)	of Secured Party	authorizing this Continuation	n Statement
P	PARTY INFORMATION CHANGE:					
	K ONE OF THESE TWO DOXES.	AND Check one of these thr — CHANGE name an		DD name: Comple	ete itemDELETE name:	Give record n
-	Change affects Debtor or Secured Party of record	item 6a or 6b; and	tem 7a or 7b <u>and</u> item 7c7	a or 7b, and item 7	c to be deleted in it	tem 6a or 6b
_	RRENT RECORD INFORMATION: Complete for Party In ORGANIZATION'S NAME	formation Change - provide	only one name (6a or 6b)			
6b.	INDIVIDUAL'S SURNAME	FIRST PER	RSONAL NAME	ADDITIO	NAL NAME(S)/INITIAL(S)	SUFFIX
	NGED OR ADDED INFORMATION: Complete for Assignme ORGANIZATION'S NAME	nt or Party Information Change - pr	ovide only one name (7a or 7b) (use exa	act, full name; do not o	mit, modify, or abbreviate any part of	the Debtor's na
7b.	INDIVIDUAL'S SURNAME					
_	INDIVIDUAL'S FIRST PERSONAL NAME					
1 8	INDIVIDUAL'S ADDITIONAL NAME(S)/INITIAL(S)					SUFFIX
33	HONE NO HONE YOU NOTE IN HOUSE NO HONE					
33				STATE	POSTAL CODE	COUNTRY
33 - 23	LING ADDRESS	CITY				
MAIL				DESTATE		CCICNI andia
MAIL	COLLATERAL CHANGE: Also check one of these four bo		DELETE collateral	RESTATE	covered collateral A	SSIGN colla
MAIL			DELETE collateral	RESTATE	covered collateral A	SSIGN colla
MAIL	COLLATERAL CHANGE: Also check one of these four bo		DELETE collateral	RESTATE	covered collateral A	SSIGN colla
MAIL	COLLATERAL CHANGE: Also check one of these four bo		DELETE collateral	RESTATE	covered collateral A	SSIGN colla
MAIL	COLLATERAL CHANGE: Also check one of these four bo		DELETE collateral	RESTATE	covered collateral A	SSIGN colla
MAIL	COLLATERAL CHANGE: Also check one of these four bo		DELETE collateral	RESTATE	covered collateral A	SSIGN colla
MAIL	COLLATERAL CHANGE: <u>Also</u> check <u>one</u> of these four bo	ixes: ADD collateral				940 SECTION AT 11 APPROXI
NAM If this	COLLATERAL CHANGE: Also check one of these four bondicate collateral:  ME OF SECURED PARTY OF RECORD AUTHORIS is an Amendment authorized by a DEBTOR, check here	ixes: ADD collateral	T: Provide only <u>one</u> name (9a o			
C In NAM If this 9a.	COLLATERAL CHANGE: Also check one of these four boundicate collateral:  ME OF SECURED PARTY OF RECORD AUTHORIS is an Amendment authorized by a DEBTOR, check here CORGANIZATION'S NAME	ZING THIS AMENDMEN	T: Provide only <u>one</u> name (9a o			SSIGN colla
. NAM	COLLATERAL CHANGE: Also check one of these four bondicate collateral:  ME OF SECURED PARTY OF RECORD AUTHORIS is an Amendment authorized by a DEBTOR, check here	ZING THIS AMENDMEN and provide name of auth	T: Provide only <u>one</u> name (9a o	or 9b) (name of As		

A. I	NAME & PHONE OF CONTACT AT FILER (optional)		7			
3. E	-MAIL CONTACT AT FILER (optional)		-			
2.5	SEND ACKNOWLEDGMENT TO: (Name and Address)		_			
٢	_	-	L			
	_:	_	THE ABOVE	SPACE IS FO	OR FILING OFFICE USE (	ONI Y
	NITIAL FINANCING STATEMENT FILE NUMBER 153388443 filed on 08/05/2015		1b. This FINANCING ST. (or recorded) in the F	ATEMENT AM REAL ESTATE	ENDMENT is to be filed [for	record]
1	TERMINATION: Effectiveness of the Financing Statement in Statement	dentified above is terminate				
	ASSIGNMENT (full or partial): Provide name of Assignee in			me of Assigno	r in item 9	
	CONTINUATION: Effectiveness of the Financing Statemen	nt identified above with respo		Secured Party	authorizing this Continuation	n Statement
Г	continued for the additional period provided by applicable law	1				
CH	<b>-</b> 100 to	ND Check one of these three				
	is Change affects Debtor or Secured Party of record	CHANGE name and/o item 6a or 6b; and ite	or address: Complete m 7a or 7b and item 7c 7a o	name: Compl or 7b, and item	ete item DELETE name: for be deleted in it	Give record n em 6a or 6b
-	URRENT RECORD INFORMATION: Complete for Party Info					
	6a. ORGANIZATION'S NAME					
3	6b. INDIVIDUAL'S SURNAME	FIRST PERS	ONAL NAME	ADDITIO	NAL NAME(S)/INITIAL(S)	SUFFIX
C	HANGED OR ADDED INFORMATION: Complete for Assignment	t or Party Information Change - provi	de only <u>one</u> name (7a or 7b) (use exact,	full name; do not o	mit, modify, or abbreviate any part of	the Debtor's na
	7a. ORGANIZATION'S NAME					
2	7b. INDIVIDUAL'S SURNAME					
	INDIVIDUAL'S FIRST PERSONAL NAME					
-						
55						SUFFIX
	INDIVIDUAL'S ADDITIONAL NAME(S)/INITIAL(S)				DOCTAL CODE	COUNTRY
	INDIVIDUAL'S ADDITIONAL NAME(S)/INITIAL(S)  MAILING ADDRESS	CITY		STATE	POSTAL CODE	
	MAILING ADDRESS		DELETE CHICAGO			
	MAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe		DELETE collateral			
	MAILING ADDRESS		DELETE collateral			
	MAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe		DELETE collateral			
	MAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe		DELETE collateral			
	MAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe		DELETE collateral			
N	MAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe		DELETE collateral			
N	MAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe		DELETE collateral			
	AAILING ADDRESS  COLLATERAL CHANGE: Also check one of these four boxe Indicate collateral:	es: ADD collateral		RESTATE	covered collateral A	SSIGN colla
N	AME OF SECURED PARTY OF RECORD AUTHORIZI	es: ADD collateral	Provide only <u>one</u> name (9a or	RESTATE	covered collateral A	SSIGN colla
N If	AME OF SECURED PARTY OF RECORD AUTHORIZ this is an Amendment authorized by a DEBTOR, check here	es: ADD collateral  ING THIS AMENDMENT: and provide name of author	Provide only <u>one</u> name (9a or	RESTATE	covered collateral A	SSIGN colla
. N	AME OF SECURED PARTY OF RECORD AUTHORIZI	es: ADD collateral  ING THIS AMENDMENT: and provide name of author	: Provide only <u>one</u> name (9a or sizing Debtor	RESTATE	covered collateral A	SSIGN colla

A. NAME & PHO	NE OF CONTACT AT FILER (optional)		7			
B. E-MAIL CONT.	ACT AT FILER (optional)		-			
C. SEND ACKNO	WLEDGMENT TO: (Name and Addres	es)	_			
	,		É			
L			THE ABOVE	SPACE IS FO	OR FILING OFFICE USE (	ONLY
	ING STATEMENT FILE NUMBER 18 filed on 08/05/2015		(or recorded) in the f	REAL ESTATE	ENDMENT is to be filed [for RECORDS rm UCC3Ad) and provide Debto	
✓ TERMINAT Statement	ION: Effectiveness of the Financing Statem	ent identified above is terminated				
	ENT (full or partial): Provide name of Assign			me of Assigno	r in item 9	
CONTINUA	TION: Effectiveness of the Financing State the additional period provided by applicable	ement identified above with respe		Secured Party	authorizing this Continuation	on Statement
-	ORMATION CHANGE:	est entered to				
Check one of thes	se two boxes:	AND Check one of these three CHANGE name and/o		O name: Comple	ete itemDELETE name:	Give record n
This Change affect	NAMES OF TAXABLE PARTY OF TAXABLE PARTY.	item 6a or 6b; and item	n 7a or 7b <u>and</u> item 7c7a o	or 7b, and item 7	c to be deleted in it	tem 6a or 6b
6a. ORGANIZA	CORD INFORMATION: Complete for Part TION'S NAME	y Information Change - provide onl	ly <u>one</u> name (6a or 6b)			
R St. INDIVIDUAL						
6b. INDIVIDUAL	-S SURNAME	FIRST PERSO	ONAL NAME	ADDITIO	NAL NAME(S)/INITIAL(S)	SUFFIX
CHANGED OR 7a. ORGANIZA	ADDED INFORMATION: Complete for AssignTION'S NAME	nment or Party Information Change - provide	le only <u>one</u> name (7a or 7b) (use exact,	full name; do not o	mit, modify, or abbreviate any part of	the Debtor's na
7b. INDIVIDUAL	L'S SURNAME					
INDIVIDUA	L'S FIRST PERSONAL NAME					
INDIVIDUAL						SUFFIX
	'S ADDITIONAL NAME(S)/INITIAL(S)				Incorn cons	COUNTRY
INDIVIDUAL		CITY		STATE	POSTAL CODE	1575727233350
INDIVIDUAL	ESS		DELETE collateral			
INDIVIDUAL	ESS AL CHANGE: <u>Also</u> check <u>one</u> of these four		DELETE collateral			
MAILING ADDRE	ESS AL CHANGE: <u>Also</u> check <u>one</u> of these four		DELETE collateral			
INDIVIDUAL  MAILING ADDRE	ESS AL CHANGE: <u>Also</u> check <u>one</u> of these four		DELETE collateral			
INDIVIDUAL  MAILING ADDRE	ESS AL CHANGE: <u>Also</u> check <u>one</u> of these four		DELETE collateral			
INDIVIDUAL  MAILING ADDRE	ESS AL CHANGE: <u>Also</u> check <u>one</u> of these four		DELETE collateral			
INDIVIDUAL  MAILING ADDRE	ESS AL CHANGE: <u>Also</u> check <u>one</u> of these four teral:	r boxes: ADD collateral		RESTATE	covered collateral A	SSIGN colla
MAILING ADDRE	ESS  AL CHANGE: Also check one of these four teral:  CURED PARTY OF RECORD AUTHOrdment authorized by a DEBTOR, check here	r boxes: ADD collateral  DRIZING THIS AMENDMENT:	Provide only <u>one</u> name (9a or	RESTATE	covered collateral A	SSIGN colla
INDIVIDUAL  C. MAILING ADDRE  COLLATER  Indicate colla  NAME OF SEC  If this is an Amenia	ESS  CAL CHANGE: Also check one of these four teral:  CURED PARTY OF RECORD AUTHOrd diment authorized by a DEBTOR, check here TION'S NAME	PRIZING THIS AMENDMENT:	Provide only <u>one</u> name (9a or	RESTATE	covered collateral A	SSIGN colla
INDIVIDUAL  C. MAILING ADDRE  COLLATER  Indicate colla  NAME OF SEC  If this is an Amenia	ESS  CAL CHANGE: Also check one of these four teral:  CURED PARTY OF RECORD AUTHOrdinent authorized by a DEBTOR, check here TION'S NAME  k National Association, as C	PRIZING THIS AMENDMENT:	Provide only <u>one</u> name (9a or zing Debtor	RESTATE	covered collateral A	SSIGN colla